

THE DETERMINANTS ON MEDICAL AND HEALTH INSURANCE OF PRUDENTIAL INSURANCE COMPANY DEMAND AMONG PEOPLE IN KLANG

NAME	MATRIC NO
MUHAMMAD FADZLI SHAH BIN CHE	08DIN18F1047
MAT FAUZI	
SYAMIM FIKRI BIN YUSRI	08DIN18F1036
MUHAMMAD LUQMANUL HAKKIM BIN	08DIN18F1033
ABD GHANI	
AMIR SHAN BIN ABDULLAH	08DIN18F1032
IMAN AIDIT HAIKAL BIN JASNI	08DIN18F1042

DIPLOMA IN INSURANCE

DEPARTMENT OF COMMERCE

NOVEMBER 2020

DECLARTION OF ORIGINALITY

We hereby declare that:

- (1) This research project is the end result of our own work and that due acknowledgement has been given in the references to ALL sources of information be they printed, electronic, or personal.
- (2) No portion of this research project has been submitted in support of any application for any other degree or qualification of this or any other university, or other institutes of learning.
- (3) Equal contribution has been made by each group member in completing the research project.

	Name of Student:	Student ID:	Signature:
1.	MUHAMMAD FADZLI SHAH BIN CHE MAT FAUZI	08DIN18F1047	
2.	SYAMIM FIKRI BIN YUSRI	08DIN18F1036	
3.	MUHAMMAD LUQMANUL HAKIM BIN ABD GHANI	08DIN18F1033	
4.	AMIR SHAH BIN ABDULLAH	08DIN18F1032	
5.	IMAN AIDIT HAIKAL BIN JASNI	08DIN18F1042	

Date: 15 DECEMBER 2020

ACKNOWLEDGEMENT

First, we would like to thank to our supervisor and lecturer of this project, Puan Maziharita and Dr Aziam Mustafa for the valuable guidance and advice. She inspired us greatly to work in this project. Her willingness in spending time on supervise us has smoothen the completion of the project.

Second, we would like acknowledge Politeknik Sultan Salahuddin Abdul Aziz Shah (PSA) for giving an opportunity for us to do this project. It also has provided a platform or database for us to gain information and resources to complete our project. In progress of completing this project, we learned the important of teamwork between group mates and it will be very useful in our career.

Third, we also want to thank our family who encouraged and fully supported us for every difficulty that comes our way.

Lastly, thanks to those respondents who was willing to spend time on answering our questionnaire. Without your cooperation, we cannot complete our task.

ABSTRACT

Medical and health insurance is a very important function to secure a family member as well as to provide security financial for individuals. However, in Malaysia the demand of medical and health insurance is yet not preferable. Not only that, medical expenses increase critically but there are still a lot of Malaysians who do not have their own medical and health insurance. Which is why conducting this research paper and it is essential to contribute ideas on the factors that affect the demand of Medical and Health Insurance. The main goal of conducting this paper is to identify the major factors that influence the Demand of Medical and Health Insurance. Relevant tests have been done base on the questionnaire survey distributed and analysed by using descriptive and inferential analysis.

Keywords: Demand, Medical and Health Insurance

TABLE OF CONTENT

Copyright Page ii
Declaration iii
Acknowledgement iv
Dedication v
Table of Contentsvi
List of Tables ix
List of Figures xi
List of Appendices xii
Preface xii
Abstract xiv
CHAPTER 1: RESEARCH OVERVIEW1
1.0 Introduction
1.1 Research Background 1
1.2 Problem Statement
1.3 Research Objectives
1.3.1 General Objective
1.3.2 Specific Objective
1.4 Research Question
1.5 Hypotheses of the Study (for quantitative research)
1.6 Significance of the Study
1.7 Chapter Layout
1.8 Conclusion
CHAPTER 2: LITERATURE REVIEW
2.0 Introduction
2.1 Review of the Literature

2.1.1 Demand on Health and Life Insurance	
2.1.2 Income Level	
2.1.3 Knowledge on Health and Life Insurance	
2.1.4 Income Protection	
2.1.5 Risk Attitude	
2.1.6 Social Influence	
2.2 Relevant Theoretical Models Review	
2.3 Proposed Theoretical/ Conceptual Framework	
2.4 Hypotheses Development	
2.5 Conclusion	
CHAPTER 3: METHODOLOGY	
3.0 Introduction	
3.1 Research Design	
3.2 The Data Collection Methods	
3.2.1 Primary Data	
3.3 Sampling Design	
3.3.1 Target population	
3.3.2 The Sampling Frame and Sampling Location	
3.3.3 The Sampling Elements	
3.3.4 Sampling Techniques	
3.3.5 Sampling Size	
3.4 Research Instrument Error! Book	mark not defined.29
3.4.1 The Questionnaire Survey	
3.4.2 The Questionnaire Design	
3.4.3 The Pilot Study	
3.5 Constructs Measurement	
3.5.2 The Ordinal Scale	
3.5.3 Interval Scale	
3.6 Data Processing	
3.6.1 Questionnaire Verification	

3.6.2 The Data Editing	
3.6.3 The Data Coding	34
3.6.4 The Data Cleaning	
3.7 Data Analysis	
3.7.1 The Descriptive Analysis	35
3.7.2 Reliability Test	
3.7.3 Normality Test	
3.7.4 Pearson Correlation	
3.7.5 Linear Regression	
3.8 Conclusion	39
CHAPTER 4: DATA ANALYSIS	40
4.0 Introduction	
4.1 Descriptive Analysis	40
4.1.1 Respondent Demographic Profile	41
4.1.2 General Information	45
4.1.3 Central Tendencies Measurement of Constructs	
4.2 Scale Measurement	55
4.2.1 Reliability Analysis	55
4.2.2 Normality Test	56
4.3 Inferential Analyses	57
4.3.1 Pearson Correlation Analysis	57
4.3.2 Simple Linear Regression	59
4.3.3 Multiple Linear Regressions	73
4.4 Conclusion	76
CHAPTER 5: DISCUSSION, CONCLUSION AND IMPLICATIONS	77
5.0 Introduction	
5.1 Summary of Statistical Analyses	77
5.2 Discussions of Major Findings	79
5.3 Implication of the study	83
5.3.1 Managerial Implications	83

5.4 Limitations of the Study	85
5.5 Recommendations for Future Research	86
5.6 Conclusion	87
References	88
APPENDIX I	

LIST OF TABLES

viii

Table 3.1: Reliability Analysis Table 3.2: Rules of thumb of the Cronbach's alpha Coefficient Size	31 36
Table 3.3: Correlation Coefficient Interpretation	38
Table 4.1: Frequency distribution for respondent's gender	41
Table 4.2: Frequency distribution of respondent's age	42
Table 4.3: Frequency distribution of respondent's geographical area	42
Table 4.4: The frequency distribution of respondent's monthly income	43
Table 4.5: Frequency distribution of respondent's educational level	44
Table 4.6: Frequency distribution for respondent's purchasing power of	
health and life insurance	45
Table 4.7: Frequency distribution for number of medical and health insur	ance
policies hold by respondents	46
Table 4.8: Frequency distribution for respondent's annual premium	47
Table 4.9: Income Level	49
Table 4.10: Knowledge on medical and health insurance	50
Table 4.11: Income Protection	51
Table 4.12: Risk Attitude	52
Table 4.13: Social Influence	53
Table 4.14: Demand on Medical and Health Insurance	54
Table 4.15: Reliability Analysis	55
Table 4.16: Normality Test	56
Table 4.17: Pearson Correlation Analysis	57
Table 4.18: Model Summary of Equation 1	59
Table 4.19: ANOVA ^b of Equation 1	59

Table 4.20: Coefficients^a of Equation 1

Page

Table 4.21: Model Summary of Equation 2	62
Table 4.22: ANOVA ^b of Equation 2	62
Table 4.23: Coefficients ^a of Equation 2	63
Table 4.24: Model Summary of Equation 3	65
Table 4.25: ANOVA ^b of Equation 3	65
Table 4.26: Coefficients ^a of Equation 3	66
Table 4.27: Model Summary of Equation 4	68
Table 4.28: ANOVA ^b of Equation 4	68
Table 4.29: Coefficients ^a of Equation 4	69
Table 4.30: Model Summary of Equation 5	71
Table 4.31: ANOVA ^b of Equation 5	71
Table 4.32: Coefficients ^a of Equation 5	72

LIST OF FIGURES

Page

Figure 2.1: The model of demand of health and life insurance	21
Figure 2.2: Proposed theoretical framework	22
Figure 4.1: Frequency for different types of insurance holding among	
respondents	48

LIST OF APPENDICES

Page

Appendix I: Questionnaires

96

CHAPTER 1 INTRODUCTION

1.0 Introduction

Chapter 1 consider as an introductory chapter for this whole study. Therefore, it covers on the research background, problem statement, research objective, and research question, hypothesis of the study, significant of the study and chapter layout. The research has comprised five independent variables and one dependent variable which are demand on medical and health insurance.

1.1 Research Background

Medical and health insurance are an insurance coverage that pays out a certain amount of money to the insured upon a certain event such as medicine and treatment. Medical insurance is insurance taken out to cover the cost of medical care. A Medical and Health Insurance (MHI) policy is generally designed to cover the cost of private medical treatment, such as the cost of hospitalization and healthcare services, if you are diagnosed with covered illnesses or have had an accident. Besides, it supports for medical expenses once individual admit to medical treatment. Health insurance is a type of insurance coverage that typically pays for medical, surgical, prescription drug and sometimes dental expenses incurred by the insured. Health insurance can reimburse the insured for expenses incurred from illness or injury, or pay the care provider directly. This insurance provides individuals, families and communities to manage income risk. According to Bank Negara Malaysia Annual Insurance Statistics (2010), in the past 10years, Malaysians started conscious about the importance of insurance. This is due to insurance consumption in Malaysia which is year 2000 to year 2010 has increased by 128% which is from RM338 to RM771 by using insurance consumption per capital.

Moreover, in the past 10 years also showed new medical and health insurance policies has also increased the number of 1,174,517 policies to 1,428,280 policies which is around 21%. Although the insurance sector has raised sharply in Malaysia, the number of medical and health insurance still consider low based on penetration rate. Besides, compare to other Asian countries, the GDP for medical and health insurance in Malaysia is low at 2.9% which conclude there is large untapped life insurance market in Malaysia. While compare to developed countries, the medical and health insurance is well appreciated by consumers due to have a higher chance to get higher education. According to Sarwar and Qureshi (2013), lack of insurance knowledge is one of the most important barriers in purchasing health and life insurance.

In past few decades, insurance is not very famous in Malaysia because the misunderstood of insurance and only small number of Malaysian bought insurance at last time. They feel insurance is an intangible product and it is only a paper with a promise and it is not required. Sometimes people know they need to buy insurance protection, but do not trust that the insurance company will fairly pay claims and they do not believe the insurance can help them in life. Some people decline the known risk, they believe that the risk would not happen to them, although they are clearly understood risks but they ignore it. Besides, some people do not bought insurance in past because of uneducated on their future financial risk and lack of the funds.

In Malaysia, public health care services are experiencing establishment pressures. This due to increase of demand and limited resources provided. Current services provide almost fully insured to the overall population and all civil servants and causes the health service suffers from overcrowding, understaffing, lack of quality and waste a lot of time in waiting. Consequently, nearly 60% of Malaysians seek private primary care (Dyah Pitaloka & Rizal, 2006). While 73.2 % of the Malaysians seek private primary care are out-of-pocket which is pay by themselves, while only 18.8% of adult Malaysians are protected. Besides, by voluntary private health insurance (VPHI) (Yu, Whynes, & Sach, 2008). They willing to pay by themselves or protected by voluntary private health insurance are often depend on their perspective on insurance.

Moreover, there are number of medical and health insurance option which pools the risks and exempt to pay any medical compensate usually correlative with out-of-pocket payment.

Besides that, National Health Insurance (NHI) is also a type of insurance which is a compulsory insurance that provide by government. However, there are a lot of disadvantages of NHI are as followed: i) they are typically not flexible which depend on legislation and political expediency for their adjustment ii) high administrative costs due to inefficiencies structures: iii) increase the burdens to government when social insurance plan is inefficient. In conclusion, the disadvantage of public and private insurance influence the public perspective whether to buy private medical and health insurance or not.

The literature suggests that there are number of factors that are important in determining the purchase behaviour on medical and health insurance. The predictions of purchase behaviour can be explained by socioeconomic and demographic variables. Both socioeconomic and demographic variables stipulate a relationship between consumption on product and service. However, some researchers dissatisfied with the accuracy of predictions of purchase behaviour based on socioeconomic and demographic variable. According to Day, Gan, Gendall and Esslemont (1991), there are various factors such as intention and attitude are potentially significant in explaining the individual purchase behaviour besides using socioeconomic and demographic variables. People who have intention to provide for those who will go on living without them must ensure more than their material well-being. And medical and health insurance is a way that enables people to deal with the ultimate loss in their time and on their terms. If failure to plan adequately for financial, consequence could bring hardship to your love one or your dependent.

1.2 Problem Statement

Medical and Health Insurance is a measure of financial security and also part of financial planning. This due to earning capacity may be ended abruptly due to old age, sickness or accident that may result in disability (permanent or temporary). Throughout the 21th century, medical and health insurance in Malaysia has growth rapidly. However, there are majority of Malaysian who do not possess a medical and health insurance policy. The number of medical and health insurance holder are still left behind compare to other Asian country.

The NHMS 2019 also reported only 22% of the population are insured with personal health insurance (MHI), with 36% of the uninsured population claiming that MHI is not necessary and a staggering 43% of them unable to afford MHI. As much as 45.5% of the Malaysian population, including about 71% of the poorest 20%, do not have any means of supplementary financial coverage for medical treatment, other than the existing tax-funded health care coverage provided by the government.

According to the statistic in year 2011, the GDP for medical and health insurance in Malaysia is low at 2.9% which this percentage included Takaful Insurance as well stated in Loke & Goh, 2012 research. Compare to other Asian countries like for example medical and health insurance GDP of Singapore is 6.1% while Japan occupied 7.5% which higher than Malaysia 3.2% and 4.6% respectively. Moreover, the statistic shown in Central Bank of Malaysia (BNM) 2013 indicated in year 2012 RM1.02 trillion sums insures was covered by Malaysian population and it is higher than the year in 2011 of RM946 billion which is increase 8%. However, this is still a large untapped medical and health insurance in Malaysia. Through the research, this paper is to study what factors influenced individuals to take up medical and health insurance as their financial protection. Besides, can define clearly the reasons Malaysian preserve their risk dropped behind critically compare to other countries.

Moreover, medical and health insurance is a very important function to secure a family member as well as to provide security financial for individuals. If the premature death happened in the breadwinner, dire the children of losing the main stream income provider. (Redzuan, 2011). Besides, medical and health insurance may hedge against the financial uncertainty which may alleviate the financial risks face by individuals. However, in Malaysia the demand of medical and health insurance is yet not preferable. Therefore, this research is going to examines what has actually influence the demand of medical and health for Malaysians and people of Klang, Selangor as a place of study.

Next, it is not a secret that for the past few years, health care costs like prescription drugs, dental care, vision care and other health-related items have been spiralling out of control and of course it had led to an increase of medical expense as well. Medical expenses are expected

to continue rising in the years ahead and this is confirmed not a joke or rumour, because a lot of surveys had been done and almost all of them predicted that the medical costs will rise as much as fifteen percent annually. Medical expenses are affected to everyone as what Chin (2014) mentioned, medical inflation is currently at 10% yearly and predicted to be rising. Next, taking a reference of Korea, Lee (2013) defined that the total hospitalization treatment and payment of depression patients increased by 21.76 percent from 12.584 billion KRW in 2004 to 15.322 billion KRW in 2007. This investigation research had resulted that human body is more vulnerable and prone to disease compare to last time.

Due to the current Covid-19 pandemic, medical and health insurance is very important to the community to cover the uncertain medical costs. We will never when we get ill or injury and the increasing of medical expenses during this pandemic.

The purpose of doing this research is because although the medical expenses is unaffordable by majority but there are still a lot of Malaysians who do not have their own medical and health insurance. This phenomenon is not healthy as it showed that most of the Malaysians do not know how to perform their own financial planning and this will lead to the financial problem. While for the next research that created by Blumber, Clemans-Cope, and Blavin (2005) resulted that the medical expenses to cure the seriously illness had accounted for bankruptcies. These researches had showed that health and life insurance is a must plan while they are managing their financial plan and worst come to worst will resulted on bankruptcy. Refer to Yong (2014), retrieved from AKPK statistic 26% credit bankruptcy rate is occupied by medical expenses in Malaysia.

1.3 Research Objectives

Objectives will be set according to the problem statement. Therefore, the research objectives must clearly determine the aim of the research which includes the general objective and specific objective.

1.3.1 General Objective

The main objective of this study is to explore the relationship between perception on medical and health insurance purchasing decision with income level, knowledge on medical and health insurance, income protection, risk attitude and social influence.

1.3.2 Specific Objective

- To identify income level will influence individual demand on medical and health insurance.
- To identify knowledge will influence individual demand on medical and health insurance.
- To identify income protection will influence individual demand on medical and health insurance.
- To identify risk attitude will influence individual demand on medical and health insurance.

[•] To identify social influence will influence individual demand on medical and health insurance.

1.4 Research Question

• Does Income is a valid predictor of determinants on demand of medical and health insurance in Klang?

• Does Knowledge is a valid predictor of determinants on demand of medical and health insurance in Klang?

• Does Income Protection is a valid predictor of determinants on demand of medical and health insurance in Klang?

• Does Risk Attitude is a valid predictor of determinants on demand of medical and health insurance in Klang?

• Does Social Influence is a valid predictor of determinants on demand of medical and health insurance in Klang?

1.5 Hypotheses of the Study

H0: Income Level has no significant relationship on demand of medical and health and life insurance in Klang.

H1: Income Level has significant relationship on demand of medical and health and life insurance in Klang.

H0: Knowledge of health and life insurance has no significant relationship on medical and demand of health and life insurance in Klang.

H1: Knowledge of health and life insurance has significant relationship on demand of medical and health and life insurance in Klang.

H0: Income Protection has no significant relationship on demand of medical and health and life insurance in Klang.

H1: Income Protection has significant relationship on demand of medical and health and life insurance in Klang.

H0: Risk Attitude has no significant relationship on demand of medical and health and life insurance in Klang.

H1: Risk Attitude has significant relationship on demand of medical and health and life insurance in Klang.

H0: Social Influence has no significant relationship on demand of medical and health and life insurance in Klang.

H1: Social Influence has significant relationship on demand of medical and health and life insurance in Klang.

1.6 Significance of the Study

The purpose of this study is to examine the factors that influence people in Klang demand of medical and health insurance. This research will identify individual may focus on which elements that may affect them on the demand. While, through this research it will be a guideline or a manual for financial institution (bank) and insurance company to let them conscious about an individual perception on demand of insurance. Other than that, this project may become a hint for a bank as well as insurance company to implement more preferable products or strategies to cope with individual mandate. While, the institutions may take into the consideration of the significant factors which may eventually help them on the profit generations and their business performance.

In addition, this research is searching for the elements which have influence the demand of medical and health insurance purchase. Furthermore, it also examines the relationship between them. After the research been done, if all of the factors are in significant it may help bank and insurance company to grow stronger cause it possible to become an antidote for them to stay competitive, while to better understand on what makes individual consider before purchasing although medical cost has increase critically.

Moreover, this thesis can provide an experience and information for future researcher, professional or even a student who are already taken part into insurance sector or even for those who wish to join in future. This study has gathered the material from other researcher, therefore who might interest to do on the similar area can takes as a source of reference.

In conclusion, this study may provide new information and contribute several materials to cope with the limitation of past study and find out the factors on demand of medical and health insurance in Klang.

1.7 Chapter Layout

First of all, five sections have been structures out for this study. Chapter 1 discusses about the introduction, research background, problem statement, research objectives, research questions, hypotheses of the study and significance of the study. In the next section, chapter 2 provides an overview of literature on perspective towards medical and health insurance. The methodology of this study in testing the effect of perspective on medical and health insurance is explained in chapter 3. Moreover, chapter 4 presents the description of patterns and analyses of the data, while chapter 5 reports the discussion, conclusion and implications of the study.

1.8 Conclusion

In this research is to examine perspective toward medical and health insurance among people in Klang. In this study is using income level, knowledge on medical and health insurance, income protection, risk attitude and social influence as independent variable to test this research.

CHAPTER 2

LITERATURE REVIEW

2.0 Introduction

Examine the factors that influenced purchasing behaviours of individual on medical and health insurance is implemented in this chapter. Later will discuss the previous researcher's point of view on the factors that will studies in this research. Besides, discovered the relationship between the demand on medical and health insurance and the determinants with framework or theoretical model in order to proposed a conceptual framework. The independent variables that have been included are income level, knowledge of health and life insurance, risk attitude, income protection and social influence. Lastly, hypotheses will be carried out in the context after the reviews.

2.1 Review of the Literature

2.1.1 Demand on Medical and Health Insurance

Health insurance plays an important role in health care financing system developing countries (Pauly, Zweifel, Scheffler, Prekar and Bassett, 2006). In Nyman (2014) mentioned demand on health insurance can be explained by two theories. First, conventional theory holds that purchase of health insurance among individual is due to they prefer certain losses to uncertain ones of the same expected magnitude. Second, the alternative theory treat health insurance is a purchased because consumers desire a transfer of income when ill.

Hwang and Geenford (2005) stated per capita premium expenditure are used to measure the demand of medical insurance. While, Browne and Kim (1993) have the similar concept with the previous researcher. Arguments has arisen by using premium income as a proxy for medical insurance consumption this due to premiums means the total revenues equal to the price multiply output (Yuengert,1993; Cummins Tennyson and Weiss, 1999) Benefits paid to policyholders could have been used as an alternative to measure medical insurance demand. This study acknowledges the existence of the problem incurred using premium income to represent the quantity of medical insurance demanded.

2.1.2 Income Level

In this era, the demand of purchasing medical and health insurances are mostly affected by the employee's salary. There are two types of employees, one is blue collar and the other one is white collar. Normally, white collar employees will have the higher salary compare to the blue-collar employees and hence they will have more intention to purchase medical and health insurance compare with the blue collar. This had been proven by the researcher, Burnett and Palmer (1984). They stated that one of the factors that will affect the demand for medical and health insurance is the customer's income. In addition, the researcher Pliska and Ye (2007) have the same point of view with demand of purchasing medical and health insurance will significantly affected on how wealthy are the wage earners. Again, look at the example of blue collar and white collar. White collar will have a stable job and stable income but for the blue collar they do not know their future income because their income is paid hourly. If they are not able to work for that particular days and their income for that month will be lesser but this is not for the white collar, they can call for medical leave if they fall in sick. Eventually white collars will have a better and stable income and will have more intention to buy life insurance.

According to Liu, Gao and Rizzo (2011), the researcher found out that the medical and health insurance package and coverage levels vary across the countries as the income level are different between each other. There are several standards of coverage in medical and health insurance, but in most countries, it still covers the cost of inpatient but not for outpatient services at all or only partially covered. This is why most of the low-income wage earners might not want to buy the insurance as the insurance they purchase might not covered all type of medical expenses.

In addition to, the others research that wrote by Wang and Rosenman (2007) observed that there is direct impact between income and demand for medical and health insurance. They suggest that location had played an important role on the demand of purchasing the life insurance. Comparing these two location urban areas and rural areas, their education levels are difference as rural areas is a less developed areas and hence there will be lesser job opportunity offer compare to the developed areas. This had shown that why in rural areas their income are lower compare to the urban area and of course this had resulted that the low income wage earner has less intensive to buy the life insurance because the cost for medical and health insurance is high and in their mindset, medical and health insurance is a luxury product on rural areas as they do not know why life insurance is so important. On the other hand, in urban areas as there are having higher income and they will buy the life insurance as a protection for their future as they know the urgency having a medical and health insurance.

Last but not least, Saliba and Ventelou (2007) stated that the demand of the health insurance is affected by income. Wage earners with higher income surely are the one who are not exempted from co-payment and they had to purchase health insurance in order to get themselves protected. Wage earners with lower income are exempted from co-payment, meanings that they do not need to disburse anything for the health care and hence they are 2/3 less unlikely to purchase health insurance than the customer with higher income. The low-income wage earners are still covered by the compulsory system but in fact, only expenditures related to the 30 diseases are reimbursed at a 100% rate.

2.1.3 Knowledge on Medical and Health Insurance

Non-buyer respondents do not have medical and health insurance due to lack of knowledge about insurance. As for knowledge factors, they haven't purchased the coverage because they have not understood well about the need for medical and health insurance (Deloitte, 2011). In year of 2012 to year 2013 Prudential Research Study founded lower knowledge makes them have less confident that they will meet their financial goals and feel less prepare to make a wise financial decision, such as a decision on purchasing insurance. As a result, they may need some professional advices before making a wise decision in purchasing insurance.

According to Sarwar and Qureshi (2013), lack of insurance knowledge is one of the most important barriers in purchasing medical and health insurance, which include the respondents unaware about it, no one has suggested and not taken by friends, family and relatives. Right now, the respondent's knowledge and awareness level about insurance are extremely low even in urban area. Knowledge about insurance is statistically significant and positively related with purchasing on medical and health insurance. It is very important as it indicates building more awareness among the respondents of health insurance will have great impact on the probability of buying medical and health insurance. (Bhat & Jain, 2006)

Salthouse (2002) stated education can be direct influence on the knowledge of insurance. Amount of education is commonly positively correlated with knowledge. Ioncica, Petrescu, Ioncica & Constantinescu (2012) stated more educated individuals are more likely to purchase insurance. This enables them to have a better access in healthcare and high security for their properties such as homes, cars, and others. Higher level of knowledge and education has the higher probability of purchasing private insurance. This is because they have lower costs of information to make decision in purchasing insurance between the complex plans offered by private insurance. Apart from that, more educated people are more confident (Mcguire, n.d.). Thus, they are more confident in their insurance knowledge and able to give advice to others on the purchasing of insurance product.

2.1.4 Income Protection

Bequest is people leave their wealth to their heirs and the heirs will enjoy their inherited wealth. Usually, the larger the bequest leaves to heirs, the easier life for heirs in future life. Usually, a person who is breadwinners, they will try to leave the larger wealth to their heirs and the higher the demanding on the medical and health insurance. The previous researchers, Lewis (1989) and Berheim (1991), founded the individual who desired to leave larger inheritance will increase the demand for medical and health insurance. This due to they would like to protect their income from being used in uncertain event. Besides, the present value of consumption of the beneficiaries increases, the demand of medical and health insurance increases as well because the policy holders know the uncertain life time and have the higher bequest motive.

Next, according to Fischer (1973) the standard model of demand for medical and health insurance is assumes the breadwinner maximizes his or her expected uncertain lifetime by choosing medical and health insurance. If the breadwinner dies, the beneficiaries will receive claims and it is shown that the demand for medical and health insurance is derived from the bequest function alone. However, Lee (2012) proved a negative relationship between bequest motives and consumption on medical and health insurance. When individual bequest motive is higher, they will prefer to save instead of having medical and health insurance as a legacy.

Arun (2012) also argue about individual buy medical and health insurance in order to secure their children when premature death happened, it is an intended to protect their children education and it is indirectly affected by income level. He founded it is very strong evidence for the hypothesis that medical and health insurances are purchased by the low-income households for bequest motives.

2.1.5 Risk Attitude

Risk attitude has significant impact on purchasing decision on medical and health insurance as stated in Kruse and Ozdemir (2004). Risk can be defined as the probability or threat of quantifiable damage, injury, liability, loss, or any other negative occurrence that is caused by external or internal vulnerabilities, and that may be avoided through pre-emptive action. The study of risk attitude of individuals has become a big interest in the growing area of behavioural finance which focuses on their financial planning practices and risk management because insurance is the transfer of risk of a group or individual to another person or a company. Apart from that, individual risk attitude towards purchasing of medical and health insurance has classified into several parts. First is about their financial risk to their perception on health risk, second pertaining to exposure of safety and environment risk and third is their experience of incidence. All of the three consciousnesses are in linkage.

Initially, risk averse decision maker buys more health insurance compare to less adverse decision maker. In order to minimize their risk of losing their wealth due to some unexpected affairs occurred and may reduce risk like for example, sickness and cause by paying off their money on medical fees they attempt to purchase health insurance for protection. This statement has been stated in the research of Heo, Grable and Chatterhee (2013). There is similar concept applied by the researcher Font and Villar (2009).

Next, Halek and Eisenhauer, (2001) mentioned the education and income with risk attitude is interrelated when making decision on purchase on medical and health insurance. Higher educational level will lead to a greater awareness of necessity of medical and health insurance. In addition, health care cost is usually bothering the lower income's people and Pauly (2007) study stated that risk averse and income level have negative relationship. Uncertainty about health condition and concern on their medical fees motivates individual's voluntary to purchase of health insurance that going to delivered money when bad events occur. The researcher Stroe and Iliescu (2010) examine the positive relationship between risk averse on the safety and environment perspective with the purchase of medical and health insurance. To eliminate worry it will increase the initiative of medical and health insurance consumption. Individual would like to change the uncertainty to certainty by ensuring their medical and health by promise to get medical and health insurance in case of death.

In addition, the research from Tennyson and Yang (2014) written the life experience may influence the demand on medical and health insurance with the changing of the risk perception.

Some demand on medical and health insurance due to previously discern on informal care provided and increase their awareness on the risk having. (Courbage and Roudaut, 2008). Lastly, family member or close friend is experiencing on the incidence with no fund provided to undergo medical treatment, thus demand of medical and health insurance been affected.

2.1.6 Social Influence

Social influence includes peer effect, family, insurance agent and so on. These have been considering as one of the important factors which influence the demand on medical and health insurance among individuals.

In Ulbinate, Kucinkiene and Moullec (2013) research mentioned that social influence and medical and health insurance demand indicated positive relationship. When individual purchase insurance is because other people are doing so. This due to it has create fear to themselves when they see their friends, neighbours and family involved in imprudent uninsured disaster and start to believe the important of insurance protection then purchase medical and health insurance by then. Therefore, individual may have the similar preferences and can reduce their search cost as well. Moreover, they may feel embarrassed if they do not have protection after they learn from others. (Kunreuther & Pauly, 2005). According to Giné et al. (2008), the information about the insurance products can be disseminating through the social networks. People conducted a more intensive marketing of the insurance product in the selected places and asked the residents help to promote the insurance product. Besides that, Dercon et al. (2011) established a peer referral treatment which benefit for every subscriber who succeeds to convince another potential client to purchase for the medical and health insurance.

Individual may know the information of medical and health insurance through their friends or people around. In Liu, Sun and Zhao (2014) research, individual have a better knowledge and experience with health insurance is through word- of mouth communication and observational learning. In addition, define that rural area consider as peer groups because mostly they will live close in distance, usually they will know each other well. Therefore, rural residents share their knowledge of medical and health insurance and may drives others to purchase on it.

Medical and health insurance agent are very crucial to understand consumers' expectations and to be able to meet their standard expectations stated by Walker and Baker (2000). First, insurance agent is a representative on insurance company to interact with customers and attract them towards medical and health insurance. This has been measured towards the service quality by insurance agent. According to Ackah and Owusu (2012), they found although many people know the word insurance from their peers, they seem not to purchase insurance to protect themselves for the future unpredicted misfortunes. On the other hand, researcher Cai, Janvry & Sadoulet (2011) stated households are more likely not to buy the insurance product if they have more weak related friends and friends that do not have influential farmers and strong knowledge in insurance.



2.2 Relevant Theoretical Models Review

Figure 2.1 The model of demand on medical and health insurance

Besides that, Arun (2012) shows that policy holders buy life insurance to secure their beneficiaries and protected their income as well. He founded income protection for bequest motive is linked with policy holders' income in Sri Lanka. It has very strong evidence shows that medical and health insurances are purchased for bequest protection by the low-income households. It means a negative relationship between income level and income protection.

In Cleeton and Zellner (1993) recommend the demand of medical and health insurance is composed of the risk attitude and income level. When individual income is low enough, the risk attitude towards consumption of insurance will be considered. This may due to they afraid

of income losses. While, it presents a negative relationship between risk averse and income level.

2.3 Proposed Theoretical/ Conceptual Framework



The Determinants of Health and Life Insurance

Figure 2.2 Proposed of theoretical framework

The sketching for figure 2.2 is to propose the conceptual framework in this research. The concept is taken from the researcher Loke and Goh (2012) with five independent variables which are income level, knowledge on medical and health insurance, income protection, risk attitude and social influence with a demand on medical and health insurance as a dependent variable.

2.4 Hypotheses Development

H0: Income Level has no significant relationship on demand of medical and health in Klang.
H1: Income Level has significant relationship on demand of medical and health in Klang.
H0: Knowledge has no significant relationship on demand of medical and health in Klang.
H1: Knowledge has significant relationship on demand of medical and health in Klang.
H0: Income Protection has no significant relationship on demand of medical and health in Klang.
H1: Income Protection has significant relationship on demand of medical and health in Klang.
H1: Income Protection has significant relationship on demand of medical and health in Klang.
H1: Income Protection has significant relationship on demand of medical and health in Klang.
H1: Risk Attitude has no significant relationship on demand of medical and health in Klang.
H1: Risk Attitude has significant relationship on demand of medical and health in Klang.
H1: Risk Attitude has no significant relationship on demand of medical and health in Klang.
H1: Risk Attitude has no significant relationship on demand of medical and health in Klang.
H0: Social Influence has no significant relationship on demand of medical and health in Klang.

H1: Social Influence has significant relationship on demand of medical and health in Klang.

2.5 Conclusion

Previous literature done by previous researchers has providing us a space to be more understands on the impact of demand on medical and health insurance. In the context, the researchers examine how those independent variables are going to affect the demand on medical and health insurance. The researchers determine and explain more detail on the relationship between each independent variable and the purchasing behaviours on medical and health insurance. In chapter 3, the researchers are going to discuss the research methodology in this paper.

CHAPTER 3

RESEARCH METHODOLOGY

3.0 Introduction

This chapter is meant to introduce clearly the overall of the research methodology in order to perform a successful research study. This chapter also discuss about the term of research design, data collection methods, sampling design, operational definitions of constructs, measurement scale and methods of data analysis.

3.1 Research Design

It is a detailed outline of how a research is conducts. A research design will involve how the data is collected, what instruments to be used and the instrument can be used to analyse the data collected. Quantitative research is fitted to this research and find it more appropriate this due to the purpose of this study is to examine the factors influenced the demand on medical and health insurance among people in Klang and the relationship of income level, knowledge on medical and health insurance, income protection, risk attitude and social influence. Quantitative methods emphasize on objective measurements and numerical analysis of data collected through polls, questionnaires or surveys. Stated quantitative research focuses on gathering numerical data and generalizing it across groups of people.

This research is a descriptive research because it considers as a type of survey method. During this survey research, the respondents or participants will answer question through online by using Google form because Covid-19 pandemic it's difficult to move. Through questionnaire it can accurately explain the relationship of income level, knowledge on medical and health insurance, income protection, risk attitude and social influence factors towards demand of medical and health insurance among people in Klang.
3.2 The Data Collection Methods

This research is using primary data to conduct. Besides, the data collection process is the main modus for this study.

3.2.1 Primary Data

Primary data (Hair, Bush & Ortinau, 2006) means the materials are never published before in anywhere else and also it is the original data collected through surveys, observation as well as research. Primary data can be considered as first-hand information and using the data for the purpose that have already intended. Primary data can be collected through numerous methods, like for instance Direct Personal Observation, Indirect Oral Interviews, Mailed Questionnaire, Schedule Method and Local Agents. Face to face survey or interview is one of the examples from the direct personal observation concept. This method is provided more reliable information and although it is time consume and may be costly. Due to the respondent will more practical and honest to answering question compare by using other method.

3.3 Sampling Design

3.3.1 Target population

Population is the entire set of units for which the survey data is used to make inferential. Target populations apply for this research is from

Klang's citizen. According to Malaysia demographic profile 2010 stated overall 240,016 populations.

3.3.2 The Sampling Frame and Sampling Location

Sampling frame consider an order of people from a population, while, the people who live in a particular place for certain treatment become part of the sampling list. For this study, the sampling sizes occupied 240 sets questionnaires and Klang's citizen are targeted. Whereas, it is impossible to distribute for overall population because it is too large and therefore the sampling locations were within Klang been conducted through convenient sampling. It be focus at Kampung Jawa, Klang.

3.3.3 The Sampling Elements

Target population about the information will be obtained through sampling unit. The purpose of this research is to identify the factors have influence the demand for medical and health insurance. Since the research is to notice what affected individual purchasing decision in Klang, hence the sample target in this research to meet the objective is:

• Samples should be the residents from Kampung Jawa, Klang.

3.3.4 Sampling Techniques

The sampling technique that used in this study is a non-probability. It is referred to the samples that gathered in a process does not give every individual equal chance of being selected in a population. The reason of using non-probability in this study is due to the population in Klang is too large and unable to collect from the entire population. Therefore, choosing non-probability as sampling technique which is using personal knowledge, convenience sampling and also based on judgment to choose the elements in the large target sizes will be better in this study. It could save up time and costs due to non-probability are convenience. In addition, this research is using convenient sampling. In the research of Babbie (2001) mentioned convenient sampling is actually a non-probability sampling technique which it is convenient accessibility and proximity.

3.3.5 Sampling Size

Sampling size defined as the number of elements to be included in this research. This study involved 240 questionnaires and distributed through hand by hand to the respondent in Klang. Wright (2009) stated the larger the sample size, the more reliable and closely the data may match with the overall population.

According to Green (1991) indicated refer rules of thumbs for individual predictors n > 104 + m. Which m refers to independent variables, whereas n imply sample size. It means the sample size must exceed 109. While, for multiple correlations require n > 50 + 8m. This shown the sample size has to go beyond 90. Research of Harris (1985) has the same point of view with the previous researcher regarding the sample size in order to get almost statistically significant result and more reliable.

3.4 Research Instrument

3.4.1 The Questionnaire Survey

In this study, it is using questionnaire to do the research in order to achieve the objective of the study. A questionnaire is a tool for collecting standardized data about a particular issue from the public. It includes clear instructions with a multiple of questions by providing enough space for answers. The reason of using questionnaires because it able to link with a big number of respondents in a short period and low cost. In addition, numerous of information able to collect through face to face interview. Besides, it can conduct through face-to-face interview modus and this method creates a valid and reliable result after the data gather. (Kahn and Cannel, 1957).

3.4.2 The Questionnaire Design

First the questionnaire design consists of cover page. Besides, the content of the questionnaires been classified into four parts which are part A, part B, part C and also part D (Appendix I). For cover page it consists of a brief explanation and details of present researcher's objectives. Moreover, contents of questionnaire in part A are general information section. The personal particular of the respondents will be collected in order to use in this project.

General information which related to this research like for instance, which agency prefer, how many policies hold, annual premium and types of insurance possess also been conducted. Next, in part B contains question about demand on medical and health insurance. Part C contains questions that aim to determine what factors influenced public towards demand of medical and health insurance. Lastly, part D, are demographic profile section. Respondent 's gender, age, races, monthly income and educational level and will be collected as the demographic have the relation with this research.

3.4.3 The Pilot Study

Pilot test is an important issue in this study because it provides pre-test for questionnaire before the officially set of questionnaires are distributed. Therefore, the pilot test has to carry out when the preparatory edition of questionnaire is ready. This is to ensure the questions are set in a correct direction and avoid any misinterpretation of question. In pilot test, the group of respondents is same to target respondent. 50 sets of questionnaires are given out and reliability test is conducted to ensure the reliable of the question set.

Variables	Cronbach's Alpha	Items' Number
Demand of Medical and Health Insurance	0.773	4
Income Level	0.811	4
Knowledge of Medical and Health Insurance	0.766	4
Income Protection	0.812	5
Risk Attitude	0.732	5
Social Influence	0.868	4

Table 3.1 Reliability Analysis

From reliable test, the result is most reliable. In this research, 50 survey forms were used to run for pilot test, with 26 items included into testing. From the table above, 4 items were used to measure demand of medical and health insurance, income level, knowledge of medical and health insurance and social influence. Four of these variables produced the alpha coefficient of 0.773, 0.811, 0.766 and 0.868 respectively. Other variables which include income protection and risk attitude were measured by 5 items, with 0.812 and 0.732 alpha coefficient respectively.

3.5 Constructs Measurement

Scale is used to measure which a system or series of marks. The researcher Sekaran (2003) also says scale is defined as a tool to distinguished individuals variable to the other. In questionnaires, three type of scales are implemented which contain nominal scale, ordinal scale and interval scale.

3.5.1 Nominal Scale

It can be defined as a distinct classification which simply some placing of data into categories. It does not have any numerical like gender, religion and occupation and it sometime could be called as labels. Yes or No scale is nominal which often use in research activities and does not have any order or distance between Yes or No. In this research questionnaire section C, those questions are based on nominal scale. Code1 to code 4 will be appoint into four groups.

3.5.2 The Ordinal Scale

It means a measurement of values what is important and significant. It is a ranking and no objective distance between any two points on your subjective scale. It often measures of non-numeric concepts like happiness and satisfaction which order from the top to bottom or biggest to smallest.

3.5.3 Interval Scale

Interval scale is a numeric scale to measure the order and also the exact differences between the values. This means that the scale and be interpret differences in the distance. Besides, the interval scale can also define by metrics such as logarithms. In research, question in section B are based on interval scale which is in the form of Likert scale. The Likert scales in the question include five different points of scales: 1 (strongly disagree), 2 (disagree), 3 (neutral), 4 (agree) and 5 (strongly agree).

3.6 Data Processing

Statistical Package for Social Sciences (SPSS) version 16 is being used to interpret data after the data is completely collected by researchers. Data collected from survey need to be undergoing four steps processes which are questionnaire checking, second transcribing third data coding and the last will be data editing.

3.6.1 Questionnaire Verification

Survey answer need to be checked by researcher and endure the interviewing quality of data collected. Moreover, questionnaire checking involves eliminating unacceptable questionnaires such as incomplete answered. This process is significant to influence the whole objectives of this research, because it assists researchers to detect errors occur and be able to undergo a prompt corrective action.

3.6.2 The Data Editing

This is a process to examine the collected raw data through questionnaire survey to detect errors and omission. If detected, correct or edit it immediately to avoid any unnecessary event occurs during data processing.

3.6.3 The Data Coding

Coding can be explained as a transformation of data into a form of understands by SPSS software. Usually it will assign a code with numbers to indicate particular reaction to certain question. Besides, data record and coding have the designated place to enter. It needs to develop in every variable and questions as well. The data will be easy for interpret and research objective can be easily interpreted is the answers are well categorized

3.6.4 The Data Cleaning

It is carried out by transferring all data into computer. It is a process of detecting and correcting any inaccurate or corrupt records. In addition, data cleaning is a broad verification for the consistency as well as rectify for the missing data. Usually consistency checking is to identify the inconsistent with the logical, range exceeded and also for the extreme value

3.7 Data Analysis

Data analysis focuses on applying statistical or logical techniques to describe and evaluate data. From this study, quantitative data analysis is applied. The programs such as Microsoft excel and SPSS software is the major tools used to enter overall data input. Thus, data is analysed and hypothesis is examined. This will contribute to statistical result and it will represent in graphical forms.

3.7.1 The Descriptive Analysis

It describes data collection's main features. The objective to do this is to summarize large sets of information that obtained from the respondents. Descriptive analysis also can be defined as the process that used to determine the reaction of the responses. From SPSS, questionnaires categorized into three sections (section A, B and C). Chapter 4 will continue to discuss further about the result, this will be determined by using graphs and tables.

3.7.2 Reliability Test

Reliability test is used to measure the quality and stability of the performance. Once obtained the same result repeatedly, this can be considered as reliable. The internal consistency reliability test is applied in this research. This test is that to judge the consistency of the result across the item. Cronbach's $alpha(\alpha)$ defined that the inter-correlations among test items means the higher the alpha, the more reliable of this scale.

(Cronbach, 1951).

Table 3.2

Cronbach's alpha (α)	Internal consistency
$\alpha \leq 0.6$	poor
$0.6 < \alpha \ge 0.7$	Moderate
$0.7 < \alpha \ge 0.8$	Good
$0.8 < \alpha \ge 0.9$	Very Good
α > 0.9	Excellent

3.7.3 Normality Test

Normality tests are to define whether the data is well-modelled by normal distribution and also to compute how the data to be normally distributed. Normality is one of the very important common assumptions in statistical procedures. Furthermore, in the research of Thode (2002) mentioned skewness and kurtosis is commonly used to test normality.

In the research of Alderson and Bachman (2004) stated the value of skewness and kurtosis can be in three categories which are positive, negative and centre on zero. While, according to rules of thumb, if the number the result of skewness and kurtosis is in between -2.0 and +2.0 it indicated a reasonably normal distribution.

3.7.4 Pearson Correlation

Pearson correlation is a test used to measure the strength of a linear association and examine the relationship between a single dependent variable and one or more independent variables. The dependent variable which is medical and health insurance demand and independent variables which are income level, knowledge on medical and health insurance, income protection, risk attitude, and social influence. Pearson correlation test can be tested into two groups which are population (R) and sample (r). In this research, correlation is using sample to measure the relationship.

There can be categorized into three types of correlation which include positive correlation, negative correlation and no correlation. These correlations characterized by concerning on one variable increase what happens to the other variable. For positive correlation, it exists when one variable increase, the other variable also increases, and vice versa. Negative correlation happens when one variable increase as the other variable decrease, and vice versa. However, when changing in one variable does not tend to either increase or decrease other variable is known as no correlation. The general rule of thumb can apply to differentiate the correlations. (Ratner, n.d.)

Value	Pearson Correlation, r
+1	perfect positive linear relationship
+0.7 to +1.0	strong positive linear relationship
+0.3 to +0.7	moderate positive linear relationship
0 to +0.3	weak positive linear relationship
0	no linear relationship
0 to -0.3	weak negative linear relationship
-0.3 to -0.7	moderate negative linear relationship
-0.7 to -1.0	strong negative linear relationship
-1	perfect negative linear relationship

Table 3. 3: Correlation Coefficient Interpretation

3.7.5 Linear Regression

Regression analysis is a method for modelling the relationship between dependent variable and one or more independent variables. It consists two types of regression analysis which are simple linear regression and multiple linear regressions. The analysis is using P value <0.01. In Nuzzo (2014) indicated a common index for the strength of evidence was 0.01 and usually interpreted as 'very significant. 'Simple linear regression is a model with dependent variable and one independent variable however multiple linear regressions is a model with dependent variable and one independent variable. The form of each type regressions is:

Simple Linear Regression: Y = a + bX + u

Multiple Linear Regression: $Y = a + b_1X_1 + b_2X_2 + B_3X_3 + ... + B_tX_t + u$

Where:

Y= Dependent Variable X= Independent Variable a= Intercept b= Slope

u= Residual

3.8 Conclusion

As conclude, this chapter discusses about the research methodology adopted in this research. It solves the problems arise from the research and discussion of research questions. The purpose of this project, the methods used to conduct the hypotheses and research questions, defining the target population precisely, kind of questionnaires, techniques used and the data analysis's findings were discussed, presented and justified in this chapter. In coming chapter, chapter 4 is providing and present the pattern of data analysis results. Besides that, it is determining the possibility for testing of hypotheses and aim to examine the relationships between each variable.

CHAPTER 4

DATA ANALYSIS AND FINDINGS

4.0 Introduction

Chapter 4 is about summarizing all result by going through the steps of inspecting, analysis and discovered the useful raw data which acquire through questionnaires. The first section discusses about the descriptive analysis, second section will be scale measurement, third, inferential analyses and the last part will me summarize the whole chapter. By using SPSS 16.0 and Microsoft Excel it is able to support for the analyses. Subsequently, descriptive and influential are used to process the data. In this research, there are 240 sets of questionnaires have been distributed and collect face to face through interview.

4.1 Descriptive Analysis

Descriptive analysis discusses about the features and the pattern of a collection of information from survey by distributed questionnaire.

4.1.1

First, in this research, it consists of 6 questions of respondents' demographic profile which are gender, age, race, geographical area, monthly income and educational level.

4.1.1.1 Gender

	-	Frequency	Percent
Valid	Female	66	27.5
	Male	174	72.5
	Total	240	100.0

Table 4.1 Frequency distribution for respondent's gender

In the table 4.1 shows the frequency distribution for respondent's gender. Based on the table 4.1, the result shows male respondents are ahead than female respondents which occupy 72.5% of the total respondent. Furthermore, 27.5% of the total respondents are female respondents. Differentiate occur between male and female is due to the research is using convenience technique when doing sampling.

4.1.1.2 Age Group

		Frequency	Percent
Valid	less than 25 years	45	18.8
	25 to 35 years	96	40.0
	36 to 50 years	89	37.1
	More than 50 years	10	4.2
	Total	240	100.0

Table 4.2 Frequency distribution of respondent's age

Based on Table 4.2, 240 respondents are answering the questionnaires. The highest respondents receive are fall into the ages group of 25-35 years old. The lowest respondents are found in age group of 50 years old and above are 4.2%. Next, for the age group of 36-50 years old the total respondents received are 37.1% and for the age group that is lesser than 25 years old are about 18.8%.

4.1.1.3 Races

	Frequency	Percent
Malay	178	74.2
India	26	10.8
Chinese	36	15.0
	240	100.0

Table 4.3 The frequence	y distribution	of respond	ent's race
-------------------------	----------------	------------	------------

Table 4.3 show the highest percentage of Race is Malay which is 74.2% (178 respondents) while for India which is consists 10.8% (26 respondents). In this research, Chinese consists 15.0% of the total respondent.

4.1.1.4 The Monthly Income

		Frequency	Percent
Valid	Less than RM2,000	64	26.7
	RM2,001~RM4,000	63	26.3
	RM4,001~RM6,000	75	31.3
	RM6,001~RM8,000	23	9.6
	Above RM8,001	15	6.3
	Total	240	100.0

Table 4.4 The frequency distribution of respondent's monthly income

Table 4.4 indicated the monthly income of 240 respondents. Initially, respondents with income range between RM4,001-RM6,000 have occupied the highest of this survey which is 31.3%. The lowest are range between RM8,001 (6.3%) and above. In addition, the monthly income of less than RM2000 hold by 64 respondents (26.7%) and monthly income range between RM2,001-RM4,000 and RM6,001 RM8,000 take up by 63 respondents and 23 respondents respectively.

4.1.1.5 Educational Level

		Frequency	Percent
Valid	Primary	1	4.0
	Secondary	13	5.4
	Diploma/College	38	15.8
	Undergraduate	24	10.0
	Postgraduate	25	10.4
	Professional Qualification	139	57.9
	Total	240	100.0

Table 4.5 Frequency distribution of respondent's educational level

Table 4.5 illustrates the result for 240 respondents from different educational background. First of all, the highest respondents from the survey are about 139 (57.9%) where there are from the professional qualification. Follow by diploma or college, which occupied 38 (15.8%) respondents. Next, secondary, undergraduate and postgraduate seize 13 (5.4%), 24 (10.0%) and 25 (10.4%) respondents respectively. Last, the lowest frequency has been occupied by primary which stated 1 (4.0%) respondents only.

4.1.2 General Information

4.1.2.4 Types of Insurance Holding

Figure 4.1: Frequency for different types of insurance holding among respondents.



Source : Developed for research

Refer to figure 4.1 which shows by using bar chart, the most respondents owned the insurance is medical card, which indicates 146 respondents. This followed by the life insurance, it shows 143 respondents. The next is 74 respondents who owned critical illness and continue with 47 respondents who hold saving or annuity. The least respondents hold the insurance is education insurance, which has only 21 respondents.

4.1.2.2 How many medical and health insurance policies hold?

		Frequency	Percentage
Valid	0	33	13.8
	1	120	50.0
	2	74	30.8
	3	6	2.5
	more	7	2.9
	than 3 Total	240	100.0

Table 4.7: Frequency distribution for number of medical and health insurance policies hold by respondents

Based on table 4.7, it shows the frequency distribution for number of medical and health insurance policies hold by respondents. Half of the 240 respondents hold only one kind of medical and health insurance. 74 out of 240 respondents hold two types of medical and health insurance. Only 6 (2.5%) and 7 (2.9%) of respondents owned three types and more than three types of medical and health insurance respectively. However, 13.8% of respondents owned none of the health and life insurance.

4.1.2.3 How much is your annual premium?

		Frequency	Percent
Valid	0-1000	56	23.3
	1000-2000	90	37.5
	2000-3000	51	21.2
	3000-4000	29	12.1
	more than 4000	14	5.8
	Total	240	100.0

Table 4.8 Frequency distribution for respondent's annual premium

Table 4.8 claimed that the frequency distribution for 240 respondents of annual premium. The most is 90 (37.5%) of the respondents paid the annual premium is about RM1,000 to RM2,000, while the least is only 14 (5.8%) of the respondents paid more than RM4,000 to the insurance provider. 56 (23.3%) of the respondents paid within RM1,000, 51 (21.2%) of them paid for RM2,000 to RM3,000 and 29 (12.1%) of them paid for RM3,000 to RM4,000.

4.1.3 Central Tendencies Measurement of Constructs

4.1.3.1 Income Level on Demand of medical and health Insurance

Table 4.9 Income Level

]	Mean	Ranking
1. I will place portion of my monthly income as	medical and health	3.3292	4
insurance premium.			
2. Once income increases, I will demand more	medical and health	3.4000	2
insurance.			
3. I will buy medical and health insurance rega	cdless of how much	3.4333	1
income I earn.			
4. My primary commitment is on medical and	health insurance.	3.3458	3

Table 4.9 shows the mean and ranking of income level on demand of medical and health insurance base on our respondents selected. Based on our result, the highest mean value is 3.4333 rank number 1 which is "I will buy medical and health insurance regardless of how much income I earn". Followed by "Once income increases, I will demand more medical and health insurance" which the mean value is 3.4000. Rank number 3 with the mean value of 3.3458 is "My primary commitment is on medical and health insurance". And the lowest value 3.3292 rank number 4 is "I will place portion of my monthly income as medical and health insurance premium".

4.1.3.2 Knowledge on Demand of medical and health Insurance

Table 4.10 Knowledge on r	medical and h	ealth insurance
---------------------------	---------------	-----------------

	Mean	Ranking
1. Knowledge of insurance protection is one of the factors that	3.6042	2
influence me on health and life insurance demand.		
2. I am very confident that I have enough insurance knowledge	3.1292	3
which incentives me to purchase on health and life insurance.		
3. I am very comfortable to give advice to friends and family	3.0917	4
members on the insurance product that will suit their needs and		
is worth buying.		
4. I prefer to have a professional advice about insurance products	3.6875	1
before purchase medical and health insurance.		
		1

Table 4.10 shows the mean and the ranking of knowledge on demand of medical and health insurance base on our respondents selected. In the table 4.10 shows "I prefer to have a professional advice about insurance products before purchase medical and health insurance" has the highest mean value 3.6875 and rank number 1. While the second highest mean value 3.6042 is "Knowledge of insurance protection is one of the factors that influence me on medical and health insurance demand". Followed by "I am very confident that I have enough insurance knowledge which incentives me to purchase on medical and health insurance" has the value of 3.1292. For the" I am very comfortable to give advice to friends and family members on the insurance product that will suit their needs and is worth buying" which is rank number 4 with the value of 3.0917.

4.1.3.3 Income Protection on Demand of medical and health Insurance

Table 4.11: Income Protection

	Mean	Ranking
1. I feel that guaranteed death benefits are the critical design element.	3.4125	2
2. I am a bread winner in family.	3.1625	3
3. I will purchase insurance to avoid out of pocket health care expenditure.	3.5542	1
4. I will purchase insurance for get the best treatment and services	3.1417	4

The table 4.11 shows the mean and ranking of income protection on demand of medical and health insurance base on our respondents selected. Through the result, "I will purchase insurance to avoid out of pocket health care expenditure" has the highest mean value which is 3.5542 and rank number 1. Next, for the second highest value 3.4125 and third highest value 3.1625 are "I feel that guaranteed death benefits are the critical design element" and "I am a bread winner in family". Rank number 4 is "I will purchase insurance for get the best treatment and services" which the value is 3.1417.

4.1.3.4 Risk Attitude towards Demand on medical and health Insurance

Table 4.12 H	lisk Attitude
--------------	---------------

	Mean	Ranking
1. I am risk averse person.	3.2292	4
2. Medical and health Insurance provide a feeling of financial safety.	3.5875	1
 Pass sickness experience on me and people around will drive me to purchase on medical and health insurance. 	3.3083	3
4. Payment guaranteed will build my confidence to purchase medical and health insurance.	3.5167	2
5. My occupation will influence my demand on medical and health insurance.	3.2208	5

In the table 4.12 shows the mean and ranking of risk attitude towards demand on medical and health insurance base on our respondents selected. The first ranking "medical and health Insurance provide a feeling of financial safety" has the highest mean value 3.5875. The second ranking with the mean value 3.5167 is "medical and health Insurance provide a feeling of financial safety". Followed by "Pass sickness experience on me and people around will drive me to purchase on medical and health insurance" has the third highest value which is 3.3083. Rank number 4 and 5 with the value 3.2292 and 3.1108 are "I am risk averse person" and "My occupation will influence my demand on medical and health insurance".

4.1.3.5 Social Influence towards Demand on medical and health Insurance

Table 4.13	Social	Influence
------------	--------	-----------

	Mean	Ranking
1. The decision on purchasing medical and health insurance is	3.5250	1
influence by my family, friends and et al.		
2. I will purchase medical and health insurance when my friends	2.8958	4
purchase.		
3. Medical and health insurance demand is cause by word-of-	3.1417	2
mouth about the benefit.		
4. I buy medical and health insurance is due to the	3.1208	3
recommendation from my friends.		

In the table 4.13 shows the mean and ranking of social influence towards demand on medical and health insurance base on our respondents selected. From the result, "The decision on purchasing medical and health insurance is influence by my family, friends and et al" has the highest mean value which is 3.5250 and it ranks number 1. Rank number 2 with the value 3.1417 is "Medical and health insurance demand is cause by word-of-mouth about the benefit". Followed by "I buy medical and health insurance is due to the recommendation from my friends" with the value of 3.1208 which rank number 3. While for the lowest ranking with the value of 2.8958 is I will purchase medical and health insurance when my friends purchase".

4.1.3.6 Demand on medical and health Insurance

		Mean	Ranking
1.	I will demand on medical and health insurance.	4.1541	1
2.	I am satisfied with the number of policies I hold.	2.3083	4
3.	My current insurance coverage is more than enough to cover my demand.	2.3958	3
4.	I realize medical and health insurance is important.	3.6083	2

Table 4.14: Demand on medical and health Insurance

Table 4.14 shows the mean and ranking for the dependent variable which is demand on medical and health insurance. The highest mean value between four question is "I will demand on medical and health insurance" which is 4.1541 and rank number 1. While for the second ranking is "I realize medical and health insurance is important" with the value of 3.6083. It followed by "My current insurance coverage is more than enough to cover my demand" which has 2.3958 mean value. And the lowest value between four questions is "I am satisfied with the number of policies I hold" with the value of 2.3083.

4.2 Scale Measurement

4.2.1 Reliability Analysis

Variables	Cronbach's Alpha	Number of Items
Demand of medical and health Insurance	0.771	4
Income Level	0.809	4
Knowledge of medical and health Insurance	0.768	4
Income Protection	0.828	4
Risk Attitude	0.735	5
Social Influence	0.805	4

Table 4.15: Reliability Analysis

Cronbach's alpha is applied in this study, the purpose is to test the internal reliability for the 6 questions with 25 items.

Cronbach (1951) stated in his work that alpha coefficient reflexes reliability. Coefficient less than 0.6 indicate poor reliability, however coefficient between 0.6 and less than 0.7 consider as moderate. Score between 0.7 and less than 0.8 are good, coefficient between 0.8 to less than 0.9 are very good. It is suggested that the nearer the value of alpha coefficient the higher the reliability. Based on table 4.15, 4 items have been used in this research to measure demand of medical and health insurance, alpha coefficient produced is 0.771. 4 items is included to measure income level, alpha coefficient equal 0.809. While 0.768 alpha coefficients shown in knowledge of medical and health insurance, takes account of 4 items. Moreover, income protection measured using 4 items produces 0.828 alpha coefficients. Number of items included in risk attitude shows 0.735 alpha coefficients. Lastly, it is 4 items to measure social influence and the outcome is alpha coefficient 0.805.

Referring to Cronbach's theory, it is suggested that all items included into testing are good enough as every variable is able to produce alpha coefficient at least level of 0.7, and shows higher coefficient providing a prove that all item used in measuring constructs are stable and consistent. Thus, it is believed that the relationships among the items are capable and reliable for the analysis.

4.2.2 Normality Test

Variables	Skewness	Kurtosis	
Demand on medical and health Insurance	-0.359	0.338	
Income Level	-0.573	0.975	
Knowledge on medical and health Insurance	0.047	0.399	
Income Protection	-0.08	-0.561	
Risk Attitude	-0.362	0.358	
Social Influence	-0.203	-0.615	

Table 4.16 Normality Test

Table 4.16 has shown the normality distribution between the variables which based on the measurement of skewness and kurtosis. This is tests on whether the data is well-modelled by normal distribution and also to compute how the data to be normally distributed. In addition, in the research of Alderson and Bachman (2004) based on rules of thumb, if the number the result of skewness and kurtosis is in between -2.0 and +2.0 it indicated a reasonably normal distribution. The result above stated, skewness in the variables of Demand of Medical and Health Insurance, Income Level, Income Protection, Risk Attitude and Social Influenced with -0.359, -0.573, -0.08, -0.362, -0.203 respectively. All are in negative sign, however, Knowledge of medical and health Insurance shown 0.047 positive values.

Skewness result is in the eligible range between -2.0 and +2.0. Next, for kurtosis all the variables are in the range of -2.0 and +2.0 as well. 0.338, 0.975,

0.399, 0.561, 0.358 and -0.615 with demand of medical and health insurance, income level, knowledge on medical and health insurance, income protection, risk attitude and social influence respectively. These indicated that data is normality distributed.

4.3 Inferential Analyses

4.3.1 Pearson Correlation Analysis

	RI	RK	RP	RR	RS	RD
RI Pearson						
Correlation	1					
Sig. (2-						
tailed)						
Ν	240					
RK Pearson						
Correlation	.427**	1				
Sig. (2-	000					
tailed)						
N	240	240				
RP Pearson						
Correlation	.372**	.235**	1			
Sig. (2-	000	000				
tailed)						
N	240	240	240			
RR Pearson						
Correlation	.398**	.516**	.503**	1		
Sig. (2-	000	000	000			
tailed)						
N	240	240	240	240		
RS Pearson						
Correlation	.194**	.293**	.035**	.153**	1	
Sig. (2-	000	000	000	000		
tailed)						
N	240	240	240	240	240	
RD Pearson						
Correlation	.458**	.475**	.517**	.601**	.124**	1
Sig. (2-	000	000	000	000	000	
tailed)						
Ň	240	240	240	240	240	240

Table 4.17: Pearson Correlation Analysis

This analysis is used to measure the strength and the relationship between two variables. Result from Table 4.17 has shown strength and the relationship between a single dependent variable and every single independent variable at a 1% significance level. Demand on medical and health insurance appears to be significant and positively correlated with five independent variable which are Income Level (r=0.458), Knowledge on Medical and Health Insurance (r=0.475), Income Protection (r=0.517), Risk Attitude (r=0.601) and Social Influence (r=0.124). Refer to the value that has indicated, Risk attitude achieved the highest positively correlated with demand of medical and health insurance and income protection. Lastly, social influence presents the weakest positively relationship to the demand of medical and health insurance.

4.3.2 Simple Linear Regression

H₀: Income Level has no significant relationship on demand of medical and health insurance in Klang.

H₁: Income Level has significant relationship on demand of medical and health insurance in Klang.

Table 4.18 Model Summary of Equation 1

Model	D	D. Course A directed D. Course	Std. Error of the	
WIUdel	К	K Square	Adjusted R Square	Estimate
1	.458 ^a	.210	.206	.66392

a. Predictors: (Constant), Income Level

Table 4.19 ANOVAb ^{of} Equation 1

Model		Sum of Squares	df	Mean Square	F	Sig.
1	Regression	27.824	1	27.824	63.123	.000 ^a
	Residual	104.909	238	.441		
	Total	132.733	239			

a. Predictors: (Constant), Income Level

b. Dependent Variable: Demand on medical and health Insurance

Table 4.20 Coefficients of Equation 1

		Unstandardized		Standardized		
		Coefficients		Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	1.521	.205		7.406	.000
	Income	.465	.058	.458	7.945	.000
	Level					

a. Dependent Variable: Demand on medical and health Insurance

According to table 4.18, R-squared is 0.21 this indicates that 21% of the variation in demand on purchase medical and health insurance can be explained by income level. In addition, the correlation coefficient between dependent variable (demand on medical and health insurance) and independent variable (income level) is 0.458.

From the table 4.19, the simple regression model (F=63.123) which define that there is significant between demand on medical and health insurance and income level with p-value $(0.000)\Box 0.01$.

As shown in Table 4.20, income level is positive related to the demand on medical and health insurance. Positive value show that an increase in income

level this will lead to demand on health and life insurance increase. The regression coefficient for income level is 0.465, this indicates that demand on health and life insurance will increase by 0.465 if income level increases by RM1, while holding another variable constant.

Based on table 4.20, p value (0.000) of income level is less than 0.01, this can explain there is significant positive relationship to demand on medical and health insurance. Therefore, H_0 is rejected.

In conclusion, income level has significant positively relationship to demand on medical and health insurance.
H₀: Knowledge on medical and health Insurance has no significant relationship on demand of medical and health insurance in Klang.

H₁: Knowledge on medical and health Insurance has significant relationship on demand of medical and health insurance in Klang.

Table 4.21 Model Summary of Equation 2

Model	D	DC		Std.	Error	of	the
	R	R Square	Adjusted R Square	Estim	ate		
1	.475 ^a	.225	.222	.6573	1		

a. Predictors: (Constant), Knowledge of Health and Life Insurance

Table 4.22 ANOVA^b of Equation 2

Model		Sum of Squares	df	Mean Square	F	Sig.
1	Regression	29.904	1	29.904	69.214	.000ª
	Residual	102.829	238	.432		
	Total	132.733	239			

a. Predictors: (Constant), Knowledge of medical and health Insurance

b. Dependent Variable: Demand on medical and health Insurance

Table 4.23	Coefficients ^a	of Equat	ion 2
------------	----------------------------------	----------	-------

		Unstandardiz	zed	Standardized		
		Coefficients		Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	1.527	.196		7.799	.000
	Knowledge of medical and health Insurance	.471	.057	.475	8.320	.000

a. Dependent Variable: Demand on medical and health Insurance

According to table 4.21, tR-squared is 0.225 this indicates that 22.5% of the variation in demand on purchase medical and health insurance can be explained by knowledge of health and life insurance. In addition, the correlation coefficient between dependent variable (demand on medical and health insurance) and independent variable (knowledge of medical and health insurance) is 0.475.

From the table 4.22, the simple regression model (F=69.214) which define that there is significant between demand on medical and health insurance and knowledge of medical and health insurance with p-value $(0.000) \square 0.01$.

As shown in Table 4.23, knowledge of medical and health Insurance is positive related to the demand on health and life insurance. Positive value show that an increase in knowledge of health and life insurance this will lead to demand on medical and health insurance increase.

The regression coefficient for knowledge of medical and health insurance is 0.471, this indicates that demand on health and life insurance will increase by 0.471 if knowledge of medical and health insurance increases by 1 unit, while holding another variable constant.

Based on table 4.23, p value (0.000) of knowledge of medical and health insurance is less than 0.01, this can explain there is significant positive relationship to demand on medical and health insurance. Therefore, H_0 is rejected.

In conclusion, knowledge of medical and health insurance has significant positively relationship to demand on medical and health insurance.

H₀: Income Protection has no significant relationship on demand of medical and health insurance in Klang.

H₁: Income Protection has significant relationship on demand of medical and health insurance in Klang.

Table 4.24 Model Summary of Equation 3

Model	R	R Square	Adjusted R Square	Std. Error of the Estimate
1	.517 ^a	.267	.264	.63929

a. Predictors: (Constant), Income Protection

Table 4.25 ANOVA^b of Equation 3

Mode	ŀl	Sum of Squares	df	Mean Square	F	Sig.
1	Regression	35.465	1	35.465	86.776	.000 ^a
	Residual	97.269	238	.409		
	Total	132.733	239			

a. Predictors: (Constant), Income Protection

b. Dependent Variable: DVMEAN

Table 4.26 Coefficients^a of Equation 3

		Unstandardiz	ed	Standardized		
		Coefficients		Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	1.528	.175		8.708	.000
	Income	.477	.051	.517	9.315	.000
	Protection					

a. Dependent Variable: Demand on medical and health Insurance

According to table 4.24, the coefficient of determination (R-squared) is 0.267. It indicates 26.7% of the variation in demand on purchase medical and health insurance can be explained by income protection. In addition, the correlation coefficient between dependent variable (demand on medical and health insurance) and independent variable (income protection) is 0.517.

From the table 4.25, the simple regression model (F=86.776) which define that there is significant between demand on medical and health insurance and income protection with p-value $(0.000)\Box 0.01$.

As shown in Table 4.26, income protection is positive related to the demand on medical and health insurance. Positive value show that an increase in income protection this will lead to demand on medical and health insurance increase.

The regression coefficient for income protection is 0.477, this indicates that demand on medical and health insurance will increase by 0.477 if income protection increases by 1 unit, while holding another variable constant.

Based on table 4.26, p value (0.000) of income protection is less than 0.01, this can explain there is significant positive relationship to demand on medical and health insurance. Therefore, H_0 is rejected.

In conclusion, income protection has significant positively relationship to demand on medical and health insurance.

H₀: Risk Attitude has no significant relationship on demand of medical and health insurance in Klang.

H₁: Risk Attitude has significant relationship on demand of medical and health insurance in Klang.

Table 4.27 Model Summary of Equation 4

Model	R	R Square	Adjusted R Square	Std. Error of the Estimate
1	.601ª	.361	.359	.59687

a. Predictors: (Constant), Risk Attitude

Table 4.28 ANOVA^b of Equation 4

Mode	1	Sum of Squares	df	Mean Square	F	Sig.
1	Regression	47.945	1	47.945	134.583	.000 ^a
	Residual	84.788	238	.356		
	Total	132.733	239			

a. Predictors: (Constant), Risk Attitude

b. Dependent Variable: Demand on medical and health Insurance

Table 4.29 Coefficients^a of Equation 4

		Unstandardize	d	Standardized		
		Coefficients		Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	.937	.192		4.882	.000
	Risk Attitude	.646	.056	.601	11.601	.000

a. Dependent Variable: Demand medical and health Insurance

According to table 4.27, the coefficient of determination (R-squared) is 0.361. It indicated 36.1% of the variation of demand on purchase medical and health insurance can be explained by risk attitude. In addition, the correlation coefficient between dependent variable (demand on medical and health insurance) and independent variable (risk attitude) is 0.601.

From the table 4.28, the simple regression model (F=134.583) which define that there is significant between demand on medical and health insurance and risk attitude with p-value $(0.000)\Box 0.01$.

As shown in Table 4.29, risk attitude is positive related to the demand on medical and health insurance. Positive value show that an increase in risk attitude this will lead to demand on medical and health insurance increase. The regression coefficient for risk attitude is 0.646, this indicates that demand on

health and life insurance will increase by 0.646 if risk attitude increases by 1 unit, while holding another variable constant.

Based on table 4.29, p value (0.000) of risk attitude is less than 0.01, this can explain there is significant positive relationship to demand on medical and health insurance. Therefore, H_0 is rejected.

In conclusion, risk attitude has significant positively relationship to demand on medical and health insurance. H₀: Social Influence has no significant relationship on demand of medical and health insurance in Klang.

H₁: Social Influence has significant relationship on demand of medical and health insurance in Klang.

Table 4.30 Model Summary of Equation 5

Model	R	R Square	Adjusted R Square	Std. Error of the Estimate
1	.124ª	.015	.011	.74104

a. Predictors: (Constant), Social Influence

Table 4.31 ANOVA^b of Equation 5

Mode	1	Sum of Squares	df	Mean Square	F	Sig.
1	Regression	2.039	1	2.039	3.713	.055 ^a
	Residual	130.694	238	.549		
	Total	132.733	239			

a. Predictors: (Constant), Social Influence

b. Dependent Variable: Demand on medical and health Insurance

Table 4.32 Coefficients^a of Equation 5

		Unstandardiz	ed	Standardized		
		Coefficients		Coefficients		
Model		В	Std. Error	Beta	t	Sig.
1	(Constant)	2.778	.182		15.252	.000
	Social	.107	.055	.124	1.927	.055
	Influence					

a. Dependent Variable: Demand on medical and health Insurance

According to table 4.30, the coefficient of determination (R-squared) is 0.015. It indicates 1.5% of the variation of demand on purchase medical and health insurance can be explained by social influence. In addition, the correlation coefficient between dependent variable (demand on medical and health insurance) and independent variable (social influence) is 0.124.

From the table 4.31, the simple regression model (F=3.713) which define that there is insignificant between demand on medical and health insurance and social influence with p-value $(0.055)\Box 0.01$.

As shown in Table 4.32, social influence is positive related to the demand on medical and health insurance. Positive value show that an increase in social

influence this will lead to demand on medical and health insurance increase. The regression coefficient for social influence is 0.107, this indicates that demand on medical and health insurance will increase by 0.107 if social influence increases by 1 unit, while holding another variable constant.

Based on table 4.32, p value (0.055) of social influence is more than 0.01; this can explain there is insignificant positive relationship to demand on medical and health insurance. Therefore, H_0 is accepted.

In conclusion, social influence has insignificant positively relationship to demand on medical and health insurance.

4.3.2 Multiple Linear Regressions

Multiple regression analysis was conducted to examine the relationship between dependent variable, which is demand of medical and health insurance and five of the independent variables including income level, knowledge of medical and health insurance, income protection, risk attitude and social influence.

Equation

Demand of medical and health Insurance = 0.02 + 0.161 Income Level + 0.183

Knowledge of medical and health Insurance + 0.237 Income Protection + 0.337 Risk Attitude + 0.000 Social Influence

		Unstandardized		Standardized		
		Coefficients	Coefficients		ť	
Model		В	Std. Error	Beta	L	Sig.
1	(Constant)	.020	.234		.084	.933
	Income Level	.161	.056	.159	2.863	.005
	Knowledge of Medical and Health Insurance	.183	.058	.185	3.138	.002
	Income Protection	.237	.052	.257	4.524	.000
	Risk Attitude	.337	.067	.313	5.034	.000
	Social Influence	.000	.043	.000	003	.997
• $R = 0.694$						
• R Square = 0.481						
• Adjusted R Square = 0.470						
	• Std. Error of the Estimate $= 0.54240$					

After running some test, a summary model has been produced to give a brief idea on how the independent variables and dependent variable are related. Table shows that the coefficient of correlation (R) is equal to 0.694, which means positive relationship could be found between both dependent and independent variables. It also indicates that there are 48.1 percent of variance in demand of medical and health insurance can be explain by all five independent variables, including income level, knowledge of medical and health insurance, income protection, risk attitude and social influence. The Table also shows that not all of the independent variables have a significant relationship with demand on medical and health insurance. For the independent variable (social influence) is insignificant to affected the dependent variable (demand on medical and health insurance), where its p-value is 0.997 greater than 0.01, given 99% confidence level.

Besides that, the tables show that while holding other variables and factors constant, the regression coefficient of 0.161 for income level shows that when there is a change in income level, it will cause a marginal change of 0.161 in demand of medical and health insurance.

Following the same principal above, when the other variables and factors are fixed, regression coefficient for knowledge of medical and health insurance is 0.183 shows that when there is a change in knowledge of medical and health insurance, it will cause marginal change of 0.183 in demand of medical and health insurance.

While income protection has regression coefficient of 0.237, explains that when there is a change in income protection, a marginal change of 0.237 in demand of medical and health insurance occurs.

Besides that, the regression coefficient of 0.337 for risk attitude shows that when there is change in risk attitude, it will cause a marginal change of 0.337 in demand of medical and health insurance.

	Sum of		Mean		
Model	Squares	df	Square	F	Sig.
1 Regression	63.891	5	12.778	43.434	.000ª
Residual Total	68.842	234	.294		
	132.733	239			
a. Predictors:	(Constant),	Sociall	MEAN, I	Protection	MEAN,

Anova Test of Multiple Linear Regression

IncomeMEAN, RAttitudeMEAN

KnowledgeMEAN,

b. Dependent Variable: DVMEAN

To examine this problem, if the model constructed (show in equation) is statistically sound, ANOVA test was performed. According to result shown in table, F-value is 43.434 and p value (0.000) which significant at 1 percent. This result show that the model constructed as equation 1 is significant and statistically sound.

This model is able to fulfil and support every hypothesis in the previous chapter. It is shown in table that all independent have impact on demand of medical and health insurance.

4.4 Conclusion

Overall, this chapter is running test which include analyse and interpret the collected data. There has interpretation on Descriptive Analysis, Scale Measurement and Inferential were implemented. While, there are 5 independents variables executed and one of the variables founded to be insignificant. Therefore, in the following chapter will discuss about major findings, managerial implications, recommendations for future researcher and also the limitation of this study.

CHAPTER 5

CONCLUSION AND RECCOMENDATIONS

5.0 Introduction This chapter is discussing five major areas which are summary of statistical analyses, major findings, implication, limitation and recommendations for future research. Moreover, this chapter also provide conclusion for the entire research project and enable policy makers and future researcher to understand clearer in the research project to allow them for future research on determinant of demand for medical and health insurances.

5.1 Summary of Statistical Analyses

Demographic	Most of the respondent	Percent%
Gender	Male	72.5%
Age	25 – 35 years	40.0%
Race	Malay	74.2%
Income level	RM4001- RM6000	31.3%
Educational level	Professional Qualification	57.9%

Table 5.1

Referring on table 5.1, majority of the respondents are male and most of the respondents are from age group of 25-35 years. Besides that, more half of the respondents are Malay. Moreover, the income level between RM4,001 to RM6000 is the majority in this paper and most of their educational level is at Professional Qualification.

On the other hand, for the general information of the respondents, majority of them had purchase insurance and half of the respondents are hold only 1 insurance policy. Most of the respondents are paid annual premium between RM1,001 to RM2,000 and majority are holding medical and health insurance.

For the construction of income level, the statement of "I will buy medical and health insurance regardless of how much income I earn" produced highest mean where the statement "I will place portion of my monthly income as medical and health insurance premium" produced lowest mean.

The statement of "I prefer to have a professional advice about insurance products before purchase medical and health insurance" produced highest mean for construct of knowledge, however the statement of "I am very comfortable to give advice to friends and family members on the insurance product that will suit their needs and is worth buying" produced the lowest mean.

Besides that, for construct of income protection, the statement of "I will purchase insurance to avoid out of pocket health care expenditure" produced highest mean and the lowest mean is the statement of "I will purchase insurance for get the best treatment and services".

Furthermore, the highest mean produced from risk attitude statement is "Medical and Health Insurance provide a feeling of financial safety" but the statement of "My occupation will influence my demand on medical and health insurance" occupied the lowest mean.

For the last framework which is social influence, the statement of "The decision on purchasing medical and health insurance is influence by my family, friends and et al" produce highest mean however the lowest mean produced by the statement of "I will purchase medical and health insurance when my friends purchase".

The result indicated, overall all the variables tested are reliable. Demand of medical and health insurance appears to be significant and positively correlated

with Income Level, Knowledge of Medical and Health Insurance, Income Protection and Risk Attitude. However, the result shows social influence is insignificant relationship with demand of medical and health insurance.

5.2 Discussions of Major Findings

According to chapter 2 literature review, previous researcher founded income level, knowledge of medical and health insurance, income protection, risk attitude and social influence is positively related with demand on medical and health insurance.

H1: Income level has positively related to demand on medical and health insurance in Klang.

Firstly, the hypothesis shows that the Income level is positive and significantly affects the demand on medical and health insurance. The multiple linear regression result indicated that p value is 0.005 which lesser than significant level (0.01) so H1 is supported. Hence, it can be concluded that the finding in the literature review is consistent with the result of this research. Based on the previous researcher, Burnett and Palmer (1984) they found out that the income level is directly proportional with the demand of medical and health insurance, this means that if the income level of the medical and health insurance holder is increasing, will lead to the holder's desire to increase the demand of the medical and health insurance.

H2: Knowledge on health and life insurance has positively related to demand on medical and health in Klang.

Next, the hypothesis proven that the knowledge on medical and health insurance will directly influenced the demand on medical and health insurance positively, the multiple linear regression with the p-value is 0.002 smaller than the alpha level (0.01). Thus, H2 is accepted. This is consistent with the previous study as the researcher discovered that most people who do not purchased the medical and health insurance is because they do not understand the important of having the insurance.

H3: Income protection has positively related to demand on medical and health insurance in Klang.

Third, the hypothesis of this study is income protection is positively affected the increase of demand on medical and health insurance proven by the p-value is 0.000 which smaller than alpha level (0.01). So, H3 is supported. The hypothesis that showed above is proven by the researcher; Lewis (1989), who defined that medical and health insurance holder will increase the demand for medical and health insurance if he or she intended to leave a huge amount of money to his family if something happened unfortunately. If his family member is increasing then of course more money is required and of course he will demand for more medical and health insurance as an income protection for them if in future something bad happened and as well anyone of us can't predict our own life time and that's why medical and health insurance is much needed. H4: Risk attitude has positively related to demand on medical and health insurance in Klang.

In addition to, the hypothesis of this study is risk attitude is positively affect demand on medical and health insurance, and the p-value is 0.000 which smaller than the alpha level (0.01) and hence, H4 is supported. The result is match with the previous researcher finding. The researcher of Heo, Grable and Chatterhee (2013) stated that risk averse buyers demand for more medical and health insurance compare to others. This because of risk adverse decision maker always plan what is the risk that may cause their income or losing their wealth due to some unexpected events such as illness, hospitalization and more. As such, they buy a lot medical and health in order to minimize the risk that had been expected.

H5: Social influence has insignificant related to demand on medical and health insurance in Klang.

Last but not least, the hypothesis of this study is social influence is insignificant to affect demand on medical and health insurance, and the p-value is 0.997 which larger than alpha level (0.01) and hence H5 is not supported. The result that shown is inconsistent with the previous research that had been done by Kucinkiene and Moullec (2013), whose research mentioned that, social had greatly influence the demand of medical and health insurance. The research found out, everyone started to purchase medical and health insurance when peers, neighbours, relatives and family started to buy. Researcher believed that this phenomenon happened because when the people around started to buy insurance it will create a feared or belief to someone that bad things will happened and in order to have the protection please follow their footstep to at least minimized the risk and hence the demand of medical and health insurance increase. OECD guidelines (2013) stated that, it is very difficult to have the survey that is aligned to the research because most of the time interviewers struggle why such information is needed and how it will be used? Although explanation and the objective of this survey had been told to the respondents but still they are still struggle to give us the answer and hence causing them to give us their view is based on their recently-experienced affective rather than their normal affective state of view. The other reasons of inconsistent between previous study is due to the cultural different. In this study, the finding showed that Malaysian is more independent to make decision in the demand of medical and health insurance. They purchased it based on self-understanding and need of the medical and health insurance instead of being influence by others. Researcher Cai, Janvry & Sadoulet (2011) mentioned, individual purchased medical and health insurance is based on their needs but not because of others influences.

5.3 Implication of the study

According to this research, founded there are some factors that affect the demand for medical and health insurance. The factor that influences the most of people in Klang demand for medical and health insurance is risk attitude followed by income protection, knowledge and income level. However, social influence factor is less important to attract people in Klang to purchase medical and health insurance.

5.2.1 Managerial Implications

This research not only use to test on demand for medical and health insurance in Klang, but also use for policy maker to boost up their sale by assessing the latent of a new market and geographical area. Based on the result, risk attitude have the highest impact on demand for medical and health insurance. Therefore, this study suggested bank and insurance companies concern more on high risk occupation and area. According to the researcher of Heo, E.Grable and Chatterhee (2013), risk adverse decision maker will buys more on health insurance to minimize their risk of losing their wealth due to some unexpected affairs occurred. They main objective is want to avoid paying off all their wealth on medical fees. Then, the bank and insurance companies can start to do research on which specific areas or occupations have the high risk on their medical and health.

Based on previous researchers, Lewis (1989) and Berheim (1991), found that the present value of consumption of the beneficiaries increases, the demand of medical and health insurance increases as well because the policy holders know the uncertain life time and have the higher bequest motive. It is essential for banks and insurance companies to keep track on the interest of policy holders to better understand their beneficiaries circumstances whether the consumption is increase or not, so that the banks and insurance companies able to approach their policy holders to demand more medical and health insurance.

Moreover, policy maker can offer different package for different income level. Based on this survey, quite some of the people are wish to buy a medical and health insurance but they tend to have financial stress on seek an amount of money to purchase an insurance. Therefore, different package according to income level enable everyone to have the purchasing power to buy the insurance. This statement also supported by the researcher Pliska and Ye (2007) which mentioned that demand of purchasing medical and health insurance will significantly affected on how wealthy are the wage earned.

Based on the result, the insignificant in social influence showed where people in Klang demand for medical and health insurance are less influence by their families, friends, insurance agents as well as advertisement. It is might because of weak in financial education. Therefore, it is suggested that the banks and insurance companies to provide financial education to a group of households in a rural area selected for their strong relationship links with others, farming skills and leadership roles and depending on social networks to expand its effect on more farmers through social learning which can be an effective way to improve in insurance take-up (Cai, Janvry & Sadoulet, 2011).

5.4 Limitations of the Study

In the progress of completing this research, a lot of limitations have been identified. The first problem that faced during the research is due to Covid-19 pandemic. The survey forms were disseminated to respondents in Klang through online.

As a result, there might be unequal amount of survey forms distributed in different states. As the questionnaire was writing in the English language, some people like older people or those poor in English language might not be really understand those questions asking about.

Next, this research only applicable in Klang. However, it only can become a reference for a reader or researcher from difference country and it is capable to be use for local scholar or policy maker. This due to the research is based on the Malaysian perception and it affected by Malaysia culture.

Other than that, the result acquired from the respondents might not be accurate from obtaining the precise information. This is because of the respondents' emotional status and their time willing to contribute to answer those questions may tend to give their answer in certain directions. In addition, the respondent may rush for time and simply answers for the question.

According to Krejcie & Morgan (1970) stated for population more than 1000000, 384 sets of questionnaires is needed to increase the accuracy and reliability. Besides, more precise results

obtained and minimize probability of error. However, in this research was taking only 240 sample size which would be inferior as compared to the sample size as suggested.

Lastly, this study was conducted by examining five independent variables which included income level, knowledge of medical and health insurance, income protection, risk attitude and social influence. The data indicated that 48.1 percent of variance in demand of medical and health insurance were explained by all five independent variables. Hence, another 51.9 percent could be explained by other extra independent variables.

5.5 Recommendations for Future Research

First and foremost, through this research, there are some recommendations for future researcher. The budget of the research was in finite and time constraint as well as there is impossible to go all the places in Klang. Which this research only able to conduct in particular states throughout Klang. Future studies can actually execute the research throughout Malaysia in order to get a more accuracy result.

The Determinants on medical and health Insurance of Prudential Insurance Company Demand among people in Klang 2020, only 240 samples sizes been conducted in this research. Therefore, the study can be carrying more samples in order to get more reliable and accurate result. In addition, suggested future researcher may distribute the questionnaire by using online and telephone survey, therefore, if sample sizes is big through these method able to make the research liable.

In the next place, it is recommended to have multi-language instead of one version which is English. This due to, it is easier for interview to carry out interview with different races.

Last but not the least, future researcher can take other variables into consideration which may influence the demand of medical and health insurance. Medical cost, Insurance agent's service,

Trust-worthy of the insurance company as a suggestion for future researcher to consider on this variable that may influence the purchasing decision as well as can better understand on this topic.

5.6 Conclusion

The result of this research indicates that risk attitude, income protection, income level as well as knowledge are four important variables correlative with the demand for medical and health insurance in Klang. However, the variable social influence fails to show the expected positive sign and need for future research to verify the connection between social influence and the demand for medical and health insurance. In addition, there are also limitations in this study and hope future research to overcome the limitation which can base on suggestions or other ways.

REFERENCES

- Ackah, C., & Owusu, A. (2012). Assessing the Knowledge of and Attitude towards Insurance in Ghana. In Research Conference on Micro-Insurance.
- Ahking, W. F., Giaccotto, C. and Santerre, E. R. (2009). The Aggregate Demand For Private Health Insurance Coverage In The United States. *The Journal of Risk and Insurance*, 76(1), 133-135.
- Alderson, J.C. & Bachman, L.F. (2004). *Statistical Analyses for Language Assessment*. New York: Cambridge University Press.
- Aliaga, M. & Gunderson, B. (2005). *Interactive statistics* (3rd ed.). New Jersey: Pearson Prentice Hall
- Altman, D.G. (1980). Statistic and ethics in medical research III: How large a sample? *British Medical Journal*, 28 (1), 1336-1338
- Babbie, E. (9th ed). (2001). *The practice of learning social research*. USA Publishers Wadsworth Thomson Learning.
- Bank Negara Malaysia (2010). In Annual Insurance Statistics 2010. Bank Negara Malaysia. <u>http://www.bnm.gov.my/files/publication/dgi/en/2010/1.1.pdf</u>
- Bhat, R., & Jain, N. (2006). Factoring Affecting the Demand for Health Insurance in a Micro Insurance Scheme. Indian Institute of Management, Ahmedabad W.P. No. 2006-07-02
- Cai, J., De Janvry, A., & Sadoulet, E. (2011). Social networks and insurance take up: Evidence from a randomized experiment in china. *ILO Microinsurance Innovation Facility Research Paper*

- Chin, C. (2014, April 20). Policyholders to pay more for coverage. *The Star.* Retrieved April 20, 2014, from http://www.thestar.com.my/News/Nation/2014/04/20/Costlymedicalinsurance-Policyholders-to-pay-more-for-coverage/
- Crosby, L. A., Evans, K. R., & Jacobs, R. S. (1991). Effects of Life Insurance Price Rebating on Stimulated Sales Encounters. *Journal of Risk & Insurance*, 58(4), 583-615.
- Courbage, C., & Roudaut, N. (2008). Empirical evidence on long-term care insurance purchase in France. *The Geneva Papers on Risk and Insurance, 33, 645–658*
- Comfrey, A. L., & Lee, H. B. (1992). A First Course in Factor Analysis. Hillsdale, NJ: Lawrence Erlbaum Associates.
- Cronbach, L.J. (1951). Coefficient Alpha and the Internal Structural of Tests. *Psychometrika*, *16*(*3*), *297-334*.
- Cummins, J. D., S. Tennyson, M. A., & Weiss (1999), Consolidation and Efficiency in the US Life Insurance Industry, *Journal of Banking and Finance*, 23(2-4),325-57.
- Day, D., Gan, B., Gendall, P., & Esslemont, D. (1991). Predicitong Purchase Behavior. Marketing Bulletin, 2(3), 18-30.

Deloitte Research. (2011). *The Voice of the Life Insurance Consumer*. Retrieved from

http://www.deloitte.com/assets/DcomUnitedStates/Local%20Assets/ Docu ments/FSI/US_FSI_Life%20Survey%20POV_042012.pdf

- Dercon, S., J.W. Gunning, & A. Zeitlin (2011a), "The Demand for Insurance Under Limited Credibility: Evidence from Kenya." In International Development Conference, DIAL.
- Educational portal- What is primary data in marketing research? Definition, sources & collection (n.d.). Retrieved August, 11 2014, form

Feldstein, M.S. (1977). Quality Change and the Demand for Hospital Care.

Econometrica, 45, 1681-1702.

- Fischer, S. (1973). A life cycle model of life insurance purchases. *International Economic Review*, 14 (1), 132–152
- Font, C.J., & Villar,G. J. (2009). Risk attitude and the demand for private health insurance: The importance of 'captive preferences'. *Journal of Annals of Public Cooperative Economics*, 80(4), 499-519. DOI:

10.1111/j.14678292.2009.00396.x

Foster, A., & Rosenzweig, M. 1995. Learning by doing and learning from others:

human capital and technical change in agriculture. *Journal of Political Economy 103(6), 1176–1209.*

- Green, S. (1991). How many subjects does it take to do a regression analysis. *Multivariate Behavioral Research*, 26(3),499-510
- Giné X., R. Townsend, & J. Vickery (2008), "Patterns of Rainfall Insurance Participation in Rural India." *The World Bank Economic Review*, 22, 539566.
- Hair, J., Bush, R., & Ortinau, D. (2006). Marketing Research within A Changing Environment (3rd ed.). New York: McGraw Hill.
- Halek, M., & J.G. Eisenhauer (2001), Demography of Risk Aversion, *Journal* of Risk and Insurance, 68(1), 1-24.
- Hersch, J. (1996), Smoking, Seat Belts, and Other Risky Consumer Decisions: Differences by Gender and Race, *Managerial and Decision Economics*, 17(5): 471-481.
- Ioncica, M., Petrescu, E.C., Ioncica, D., & Constantinescu, M. (2012). The Role of Education on Consumer Behavior on the Insurance Market. *Procedia- Social and Behavioral Sciences*, 46, 4154-4158.
- Jackson, S.L. (2009). Research Methods and Statistics: A Critical Thinking Approach 3rd edition.
- Kahn, R.L. & Cannel, C. F. (1957). Technique for motivating the respondent. *In The Dynamic of Interviewing* (p. 65-90). New York: Wiley.

- Kissoon, J. (2011). Introduction to Computers and it Major Functions. Retrieved February 16, 2014, from Clever Logic: <u>http://www.cleverlogic.net/book/introduction-computers-and-it-majorfunctions</u>
- Kok, C. (2009, March 14). Rising Pressure of Healthcare Cost. The Star. Retrieved April 14, 2014, from <u>http://www.thestar.com.my/story.aspx/?file=%2f2009%2f3%2f14%2fb</u> <u>usin ess%2f3456038</u>

Korobkin, R. (2014). Comparative effectiveness research as choice architecture:

the behavioral law and economics solution to the health care cost crisis. Michigan Law Review, 112(4), 523-574.

- Krejcie, R.V., & Morgan, D.W. (1970). Determining sample size for research activities. *Educational and Psychological Measurement, 30, 607-610.*
- Kruse, J.B. & Ozdemir, O. (2004). Disaster Losses in the Developing World: Evidence from the August 1999 Earthquake in Turkey. *Turkish Economic Association Discussion Paper*.
- Krueger, B. A. & Kuziemko, I. (2011). The Demand for Health Insurance among Uninsured Americans: Results of a Survey Experiment and Implications for Policy. *Working paper*, 217.
- Kunreuther, H. C., & Pauly, M. (2005). Insurance Decision-Making and Market Behaviour. Foundations and Trends® in Microeconomics, 1(2), 63-127. http://dx.doi.org/10.1561/0700000002
- Lee, H.S (2013). The Impact of Emergency Room Utilization by Depression Patients on Medical Treatment Expense in Korea. Osong Public Health and Research Perspectives, 4(5), 240-245.
- Lee, Y. W. (2012). Asymmetric information and the demand for private health insurance in Korea. *Economics Letters*, *116*(*3*), *284-287*.
- Lee, & M.L. (2012). Incidents Bequests: Bequest motives and the choice of selfinsure late-life risks. *Review of Economic Dynamics* 15(2), 226-243.

- Lim, C. C., & Haberman, S., (2002). Macroeconomic Variables and the Demand for Life Insurance in Malaysia. Faculty of Actuarial Science and Statistics, CASS Business School, City University (London)
- Liu, Sun, & Zhao, 2014. Social learning and health insurance enrolment: Evidence from China's new cooperative medical scheme. *Journal of Economic Behavior & Organization*, 97 (C), 84-102.
- Liu, H., Gao, S. & A. Rizzo J. (2011). The Expansion of Public Health Insurance and the Demand for Private Health Insurance in Rural China. *China Economic Review*, 22(1), 28-41..
- Lynch, M.S. (2006). Explaining Life Course and Cohort Variation in the Relationship between Education and Health: The Role of Income. *Journal of Health and Social Behavior*, 47(4), 324-338. DOI: 10.1177/002214650604700402

Measuring subjective well-being (Eds). (2013). OECD Guideline.

- Mcguire, J. (n.d.). *The Benefits of Having a College Education*. Retrieved July 15, 2014, from http://www.collegeview.com/articles/article/the-benefits-of-
- Nuzzo, R. (2014). Scientific method: Statistical errors. Nature, 50(6), 150-15.
- Nyman, J.A. (2014). Demand for and welfare implications of health. Encyclopedia of Health Economics, 1(1), 159-166.
- Prudential Research Study. (2012-2013). Financial Experience & Behaviors Among Women. Retrieved from <u>https://www.cgsnet.org/ckfinder/userfiles/files/Pru_Women_Study.pdf</u>
- Pitaloka S. Dyah., & Rizal A.M. (2006). Patients' Satisfaction in Antenatal Clinic Hospital Universiti Kebangsaan Malaysia. Jurnal Kesihatan Masyarakat (Malaysia), 12 (10), 1-10
- Ratner, B. (n.d.) . The Correlation Coefficient: Definition. Retrieved April 14, 2014, from

Pauly. V.M. (2007). Risks and benefits in health care: The view from economics.

The Journal of Health Aff (Millwood), 26(3), 653-662. DOI:10.1377/hlthaff.26.3.653

- Saliba, B. & Ventelou, B. (2007). Complementary Health Insurance in France Who Pays? Why? Who will Suffer from Public Disengagement? *Health Policy*, 81(2), 166-182.
- Salthouse, T. A. (2002). Interrelations of Aging, Knowledge, and Cognitive Performance. In Understanding Human Development: Dialogues with Lifespan Psychology (p.265-p.287). US: Kluwer Academic.
- Sarwar, A., & Qureshi, H.A. (2013). Awareness and Willingness to Buy Private Health Insurance and a Look Into its Future Prospects in Pakistan, European Journal of Business and Social Sciences (EJBSS), 2(1), 69-81, ISSN: 2235-767X.

Sekaran, U. (2000). Research Methods for Business: A Skill Building Approach

(4" ed.). NY: John Wiley & Sons.

- Stroe, M.A., & I, M. (2013). Attitudes and perception in consumers's insurance decision. *Global Economic Observer*, 1(2),112-120.
- Tannahill, B. A. (2013). The Role of Insurance in Retirement Planning. *The Journal of Financial Service Proffessionals*, 67(4), 32-35.
- Thode, H.C. (2002). Testing for normality. New York: Marcel Dekker.
- Vaithianathan, R. (2006). Health insurance and imperfect competition in the health care market. *Journal of Health Economics*, 25(6), 1192-1202..
- Walker, J., & Baker, J. (2000). An exploratory study of a multi-expectation framework for services. *Journal of Services Marketing*, 14 (5), 411-431

- Wang, H. Holly. & Rosenman, R. (2007). Perceived Need and Actual Demand for Health Insurance among Rural Chinese Residents. *China Economic Review*, 18(4), 373-388.
- Wilson Van Voorhis, C. R. & Morgan, B. L. (2007). Understanding Power and Rules of Thumb for Determining Sample Sizes. Tutorials in Quantitative Methods for Psychology, 3(2), 43-50.

Wright, C. (2009). A guide to sampling & statistical reliability. Retrieved July 10,

2014,form

http://www.sandwelltrends.info/lisv2/briefingnotes/Technical_Note_5_ _A_guide_to_sampling_and_statistical_reliability.pdf

Yong, C. (2014). (2014, May 19). *Bankruptcies up to 32% 1Q. The Edge.* Retrieved June 20, 2014, from <u>http://www.theedgemalaysia.com/businessnews/290000-bankruptcies-up-32-in-1q.html</u>

Yu, C. P., Whynes, D. K., & Sach, T. H. (2008). Equity in health care financing:

The case of Malaysia. International Journal for Equity in Health, 7(15).

APPENDICES

SECTION A: GENERAL INFORMATION

1.	Which	Insurance	Company	do	vou	prefer?
т.	***	mouranee	Company	uv	J04	protor .

- AllianzPrudential
- Great Eastern
- □ ING
- \Box AIA
- □ Manulife
- □ Zurich
- □ _{Kurnia}
- □ UniAsia

Others: _	
-----------	--

2. How many health and life insurance policies hold?

- \Box 0
- \square_1
- \square_2
- \square_3
- \square more than 3
- 3. How much is your annual premium?
- **RM0-RM1000**
- □ RM1001-RM2000
- □ RM2001-RM3000
- □ _{RM3001-RM4000}
- \square RM4001 and above

- 4. Please ($\sqrt{}$) on the types of Insurance holding.
- □ Medical Card
- □ Saving/ Annuity
- Critical Illness
- □ Life Insurance
- Education Insurance
SECTION B

DEPENDANT VARIABLE

	SA	Α	Ν	D	SD
1. I will demand on medical and health insurance	5	4	3	2	1
2. I am satisfied with the number of policies I hold.	5	4	3	2	1
3. My current insurance coverage is more than enough to cover my demand.	5	4	3	2	1
4. I realize medical and health insurance is important.	5	4	3	2	1

I. Respondents Demand on Medical and Health Insurance

SECTION C

Please tick ($\sqrt{}$) the appropriate answer or fill in the details in the column provided.

Please indicate your degree of agreement on the following statements by circling

the numbers given ranging from:

Strongly Disagree=1, Disagree=2, Neutral=3, Agree=4, Strongly

Agree=5

I. Respondent's Income Level on Medical and Health Insurance

	SA	Α	N	D	SD
1. I will place portion of my monthly income as health and life insurance premium.	5	4	3	2	1
2. Once income increases, I will demand more health and life insurance.	5	4	3	2	1
3. I will buy medical and health insurance regardless of how much income I earn.	5	4	3	2	1
4. My primary commitment is on medical and health insurance.	5	4	3	2	1

II. Respondent's Knowledge on Demand of Medical and Health Insurance

	SA	Α	Ν	D	SD
1. Knowledge of insurance protection					
is one of the factors that influence me	5	4	3	2	1
on medical and health insurance					
demand.					
2. I am very confident that I have					
enough insurance knowledge which	5	4	3	2	1
incentives me to purchase on medical					
and health insurance					
3. I am very comfortable to give advice					
to friends and family members on the	5	4	3	2	1
insurance product that will suit their					
needs and is worth buying					
4. I prefer to have a professional advice					
about insurance products before	5	4	3	2	1
purchase medical and health insurance					

III. Respondent's Income Protection on Demand of Medical and Health Insurance

	SA	Α	N	D	SD
1. I feel that guaranteed my medical expenses can be paid	5	4	3	2	1
2. I am a bread winner in family.	5	4	3	2	1
3. I will purchase insurance to avoid out of pocket health care expenditure.	5	4	3	2	1
4. I will purchase insurance for get the best treatment and services	5	4	3	2	1

IV. Risk Attitude Towards Respondent's on Demand of Medical and Health Insurance

	SA	Α	Ν	D	SD
1. I am risk averse person					
	5	4	3	2	1
2. Medical and health Insurance					
provide a feeling of financial	5	4	3	2	1
safety.					
3. Pass sickness experience on me	_				
and people around will drive me to	5	4	3	2	1
purchase on medical and health					
insurance.					
4. Payment guaranteed will build	_	_			
my confidence to purchase	5	4	3	2	1
medical and health insurance.					
5. My occupation will influence					
my demand on medical and health	5	4	3	2	1
insurance					

V. Social Influence Towards Respondent's on Purchasing Decision of Medical and Health Insurance

	SA	Α	Ν	D	SD
1. The decision on purchasing medical and health is influence by my family, friends and et al.	5	4	3	2	1
2. I will purchase medical and health insurance when my friends purchase.	5	4	3	2	1
3. Medical and health insurance demand is cause by word-of-mouth about the benefit.	5	4	3	2	1
4. I buy medical and health insurance is due to the recommendation from my friends.	5	4	3	2	1

SECTION D: DEMOGRAPHIC PROFILE

Please tick ($\sqrt{}$) the appropriate answer or fill in the details in the box provided.

1. Gender:

□ _{Male}

- □ _{Female}
- 2. Age:
- \Box less than 25 years
- \Box 25 35 years
- \Box 36 50 years
- \Box 50 and above
- 3. Race:
- □ _{Malay}
- □ India
- \Box Chinese
- \Box Others

4. Monthly Income Level:

- Less than RM2,000
- □ RM2,001- RM4,000
- □ RM 4,001- RM6,000
- □ RM6,001- RM8,000
- □ Above RM8,001

- 5. What is your highest educational level?
- D Primary
- □ Secondary
- Diploma / College
- □ Undergraduate
- D Postgraduate
- D Professional Qualification (Accountant, Doctor, Pilot etc)